CFR WORKING PAPER NO. 16-05 call of puty: pesignated market maker participation in call auctions ε . Theissen \bullet c. Westheide centre for rinancial research Look deeper

Call of Duty: Designated Market Maker Participation

in Call Auctions

Erik Theissen Christian Westheide[†]

This version: August 21, 2017

Abstract

On many equity markets, designated market makers (DMMs) supply additional

liquidity for small and mid cap stocks. Whereas prior research has focused on their

role in continuous trading, we analyze their activity in call auctions. Using data from

Germany's Xetra system, we find that DMMs are most active when they can provide

the greatest benefits to the market, i.e., in relatively illiquid stocks and at times of

elevated volatility. They stabilize prices and earn positive profits. These results imply

that DMMs provide a valuable service to the market, and that they charge an implicit

price for that service.

Keywords: Designated market makers, Call auctions

JEL classification: G10

We thank Deutsche Börse for providing historical data. Financial support from Deutsche Forschungsgemeinschaft (Grant TH 724/4-1) is gratefully acknowledged. We thank Satchit Sagade, Stefan Scharnowski and Kumar Venkataraman for valuable comments. Erik Theissen, University of Mannheim and Centre for Financial Research (Cologne), e-mail theissen@uni-mannheim.de; Christian Westheide, University of

mannheim.de.

1

Mannheim, Research Center SAFE and Center for Financial Studies (Frankfurt), e-mail westheide@uni-

1 Introduction

Many equity markets operate a hybrid market structure that combines continuous trading in the electronic open limit order book during most of the trading day with call auctions at the open and the close. While liquidity voluntarily provided by market participants is usually sufficient for large cap stocks, many exchanges rely on designated market makers (DMMs) for mid and small caps. We are the first to study the activity of DMMs in call auctions in the context of a hybrid market structure. We find beneficial effects of DMM participation. Their activity is negatively related to the liquidity of the stocks and increases at times of high volatility and uncertainty. By leaning against the market, DMMs limit transitory price movements and thus stabilize prices. Their trades predict future returns and, hence, are profitable. This suggests they are implicitly compensated for their services. Thus, despite the fact that they have binding obligations such as maximum spread requirements, DMM activity in call auctions appears viable even in the absence of payments from the issuer or the exchange. In addition to the results immediately related to DMM activity, we show that the closing auction generally accounts for a much higher fraction of the total trading volume than the opening auction, and that the relative importance of the closing auction is increasing in firm size while the relative importance of the opening auction is decreasing in size.

Previous empirical evidence on designated market making is almost exclusively based on data from the continuous trading sessions. Little is known about whether and how designated market makers contribute to the liquidity of the call auctions. Call auctions form an important part of today's hybrid trading systems for at least three reasons: they attract a significant share of the daily trading volume, they serve to impound important information into prices at the open, and they define the closing price, which is increasingly relevant, e.g., for mutual funds' daily determination of their net asset values. Thus, an understanding of whether and how DMMs can improve call auction outcomes is critical.

We use data from the Xetra system operated by Deutsche Börse. Trading in Xetra opens and closes with a call auction. Further, there is a regular intraday call auction. Finally, trading is restarted with a call auction after volatility-induced trading halts. Our data identifies the trades of designated market makers. We are therefore able to analyze to which extent designated market makers participate in the auction, and to analyze the cross-sectional and time-series determinants of their trading activity.

Our paper contributes to the literatures on designated market makers and on call market trading. The former has mostly focused on DMMs in continuous limit order books. The theoretical models by Bessembinder et al. (2015) and Sabourin (2006) predict that the presence of a designated market maker increases liquidity. These theoretical predictions are supported by a growing body of empirical evidence which suggests that designated market makers indeed improve liquidity (e.g. Anand et al. (2009), Anand and Venkataraman (2016), Clark-Joseph et al. (2016), Declerck and Hazart (2002), Eldor et al. (2006), Hengelbrock (2011), Menkveld and Wang (2013), Nimalendran and Petrella (2003) and Skjeltorp and Ødegaard (2015)).

The relative advantages of electronic call auctions have been discussed theoretically by Cohen and Schwartz (1989), Economides and Schwartz (1995) and Schwartz (2000). Several papers (e.g. Pagano and Schwartz (2003), Chang et al. (2008), Chelley-Steeley (2008), Chelley-Steeley (2009), Kandel et al. (2012), Pagano et al. (2013)) provide evidence that the introduction of an opening and / or a closing call auction improves market quality, particularly for small caps.¹ Abad and Pascual (2010) and Zimmermann (2014) analyze call auctions subsequent to volatility-induced trading halts, an institutional feature that also plays a role in the present paper.

While all the above papers analyze trading in call auctions, none of them considers the role of designated market makers in these auctions. The only papers we are aware of that explicitly

¹Interest in this issue has been reignited recently when the London Stock Exchange introduced a midday call auction. Note that our setting does not allow analyzing the effect of introducing a midday call auction because such an auction was conducted for all our sample stocks during the entire sample period.

consider designated market makers in call auctions are Madhavan and Panchapagesan (2000), Kehr et al. (2001) and Venkataraman and Waisburd (2007). Madhavan and Panchapagesan (2000) and Kehr et al. (2001) analyze floor-based trading systems in which the designated market maker (the NYSE specialist and the "amtlicher Kursmakler" on the Frankfurt Stock Exchange, respectively) had exclusive access to the limit order book and had discretionary power in setting the auction price, which is entirely absent in today's electronic call markets. Venkataraman and Waisburd (2007) use data from the French equity market which operates an electronic call auction. They find that firms with designated market makers have better market quality and that share prices increase upon the announcement that a designated market maker will be introduced. Our paper differs from theirs in two important ways. First, the French stocks analyzed in Venkataraman and Waisburd (2007) are traded only in call auctions while the stocks we analyze are traded continuously. Second, the data set analyzed in Venkataraman and Waisburd (2007) does not identify trades made by designated market makers. Therefore, they can test whether the existence of a designated market maker affects market quality, but they cannot analyze the trading activity of designated market makers. In a sense, then, our results complement theirs because we show in detail how designated market makers improve the market quality in the call auctions.

The remainder of this paper is organized as follows. Section 2 describes the institutional setting. Section 3 develops our hypotheses. Section 4 describes our data set and presents descriptive statistics. Section 5 contains the empirical analysis of market maker participation, section 6 concludes.

2 The Institutional Setting

Xetra, operated by Deutsche Börse, is the dominant market for German stocks. Stocks can be traded either continuously or in a call auction-only mode. Which trading protocol applies for a particular stock is determined in a two-step procedure. In a first step the

exchange sorts stocks into two liquidity categories, A and B, according to the turnover and the transaction costs² measured over a period of four months. Stocks categorized as liquid are traded continuously. The issuer may voluntarily contract with a designated market maker.³ Stocks categorized as illiquid, on the other hand, are traded in a call auction-only mode unless they have a designated market maker. As explained in more detail below, employing a designated market maker is costly. Firms may be willing to incur these costs because a switch to continuous trading has been shown to improve market quality and to result in an increase in share prices (Amihud et al. (1997) and Kalay et al. (2002)).

From this it follows that there are four groups of stocks: (1) illiquid stocks that do not have a designated market maker and that are traded in a call auction-only mode, (2) illiquid stocks that do have a designated market maker and are traded continuously, (3) liquid stocks that have a designated market maker (on a voluntary basis) and (4) liquid stocks that do not have a designated market maker. Since the focus of this paper is the participation of designated market makers in call auctions we only include stocks from groups 2 and 3 in our sample. Consequently, all our sample stocks are traded continuously, and all sample stocks have at least one designated market maker.

The trading day in Xetra starts at 9 a.m. with an opening call auction and ends at 5:30 p.m. with a closing auction. A third call auction takes place in the middle of the trading day (between 1 p.m. and 1:17 p.m.⁴). The continuous trading session, which comprises the remainder of the trading day, is organized as an electronic open limit order book. Trade execution is governed by price and time priority. Continuous trading is halted when the

²Transaction costs are measured by the roundtrip cost of a trade of size 25,000 Euros.

³The official name of designated market makers in Xetra is "designated sponsors". We use the term designated market maker (DMM) instead, which is more common in the academic literature.

⁴The intraday call auction is held between 1:00 and 1:02 for DAX and TecDAX stocks, between 1:05 and 1:07 for MDAX and SDAX stocks, and between 1:15 and 1:17 for other stocks. The MDAX, SDAX and TecDAX are indices calculated by Deutsche Börse AG. They comprise stocks which are listed in the prime standard segment of the Frankfurt Stock Exchange and which are (as measured by free float and trading volume) smaller than the 30 stocks included in the blue-chip index DAX. The MDAX and SDAX each comprise 50 stocks from non-technology sectors. The 50 stocks in the MDAX are the next 50 stocks outside the technology sector after the DAX stocks while the 50 stocks in the SDAX are those that follow after the MDAX stocks. The TecDAX comprises the 30 largest technology stocks outside of the DAX.

price hits a predefined (but undisclosed) price limit. After such a volatility interruption trading is restarted with a call auction. There are thus four types of call auctions, namely, opening auctions, closing auctions, (regular) intraday auctions, and auctions after volatility interruptions.

Orders submitted to Xetra belong to one of three account types, "agency", "principal" or "market maker". Agency orders are submitted by Xetra members on behalf of other traders (i.e. orders submitted by Xetra members acting as brokers for their customers). Principal orders are orders submitted by Xetra members on their own behalf. Market maker orders are orders submitted by Xetra members in their capacity as designated market makers.

The designated market making arrangement is specified in a contract between the issuer and the market maker. The issuer pays the market making firm a fee. The market making firm, in turn, commits to register as a designated market maker for the issuer's stock. The minimum standards for designated market makers are defined in the "Designated Sponsor Guide" published by Deutsche Börse. They are required to submit buy and sell limit orders (referred to as "quotes" hereafter) to the call auctions and to quote bid and ask prices during the continuous trading session. They have to meet a minimum participation rate in the call auctions and a minimum quotation time in the continuous trading session. For a quote to count towards the minimum participation rate and minimum quotation time requirements, it must satisfy maximum spread and minimum depth requirements. The details of these requirements are outlined in the Designated Sponsor Guide. Designated market makers do not have an informational advantage (such as exclusive access to the limit order book, as the NYSE specialists had), and their quotes are subject to the same price and time priority rules as orders submitted by agency and principal traders.

Provided designated market makers fulfill their obligations, they benefit from a rebate on the execution fees for illiquid stocks (group 2). Deutsche Börse monitors the performance of the designated market makers and publishes a quarterly rating.⁵

⁵See Theissen and Westheide (2015) for details.

Many firms have more than one market maker. There are two reasons why a firm can have several market makers. First, the issuer can voluntarily contract with more than one market maker. Second, market making firms can register as market makers in a particular security without entering into a contract with the issuer. In this case, they have to comply with the full set of requirements for designated market makers but do not receive a fee from the issuer. We cannot differentiate between these two cases because the existence (and the terms) of a contract between the issuer and the market maker are not disclosed.

3 Hypotheses

While the liquidity of a pure order-driven market is usually sufficient for large and high-volume stocks, it often is not for small caps. In fact, the intention of integrating designated market makers into (continuous and/or call) auction markets has been to increase liquidity for stocks that are inherently illiquid (see e.g Nimalendran and Petrella (2003) and Menkveld and Wang (2013)). By this argument, we expect that the relative importance of designated market makers, measured by their share in the auction trading volume, is inversely related to firm size.

Hypothesis 1 Designated market makers account for a higher share of the total auction volume for stocks that are inherently less liquid (i.e., stocks with lower market capitalization).

As we will document in the next section, there are significant differences in trading volume between the four auction types (opening, intraday and closing auction, and auctions after volatility interruptions). By a similar argument as above, we expect the services of designated market makers to be in relatively higher demand in those auctions that attract less volume.

Hypothesis 2 The share of designated market makers in the total auction volume is higher in those auctions that attract less overall volume.

We expect this hypothesis to hold in two dimensions. First, at an aggregate level, it should hold across auction types. If, for example, opening auctions generally attract less volume than closing auctions, then we expect market maker participation to be higher in the opening auction. Second, the hypothesis should hold in the time series. For a given stock and a given auction type, designated market maker participation should be higher on days with lower auction volume.

Designated market makers have an affirmative obligation to submit buy and sell orders to the auction, and they are subject to maximum spread requirements. Principal and agency traders, on the other hand, can withdraw from the market at times of high volatility or increased informational asymmetry. We thus have the following hypothesis.

Hypothesis 3 Designated market maker participation is increasing in both return volatility and adverse selection risk.

The NYSE specialists had the affirmative obligation to stabilize prices (e.g. Hasbrouck and Sofianos (1993), Panayides (2007)). Even though the designated market makers in Xetra do not have such an affirmative obligation, they can nevertheless be expected to contribute to price stabilization. As suppliers of liquidity they will typically trade on the short side whenever there is an order imbalance. To the extent that the imbalance causes temporary price pressure, designated market makers will thus trade against the temporary price change.

Hypothesis 4 Designated market makers stabilize prices, i.e., they trade against temporary price changes.

Designated market makers provide liquidity to the market and thus provide a valuable service to other traders. Providing this service is both costly and risky. Consequently, DMM should expect a compensation. We thus have

Hypothesis 5 Designated market makers earn positive trading profits.

4 Data and Descriptive Statistics

Our data set contains all transactions executed in stocks that are traded continuously in Xetra during the months of July and August 2011 (44 trading days) and July and August 2013 (45 trading days). We exclude all stocks that do not have a designated market maker. Note that the most liquid stocks (in particular the component stocks of the DAX index) did not have a designated market maker during our sample period. Therefore, our dataset is tilted towards small and mid caps. We further exclude foreign stocks (defined as stocks with an ISIN country code different from DE) and stocks that do not have at least one call auction with non-zero volume on at least 50% of the trading days in our sample period. We match the transactions data with intraday data on best bid and ask prices provided by Deutsche Börse. We further obtain data on firm characteristics from Thomson Reuters Datastream. After the exclusion of five stocks because of missing data, the final sample contains 250 stocks in 2011 and 209 stocks in 2013.

Table 1 provides summary statistics for the stocks in our sample. Columns 1-5 [6-10] show data for the first [second] sample period. The average sample stock has a market capitalization of 689.8 [993.1] million Euros in the 2011 [2013] sample, and an average daily trading volume of 2.80 [2.15] million Euros. There is considerable cross-sectional variation in the sample. While the firm at the 5th percentile has a market capitalization of 25.4 [30.3] million Euros and an average daily volume of 11,719 [17,166] Euros, the corresponding values for the firm at the 95th percentile are 3,050.2 [4,715.8] million Euros for the market capitalization and 14.7 [9.6] million Euro for the average daily volume. The average share price is 21.68 [25.77] Euros in the 2011 [2013] sample. The average quoted bid-ask spread is 1.10% [0.74%] while the average 5-minute price impact (a proxy for losses to informed traders) is 0.54% [0.40%]. The large differences in the bid-ask spreads and price impacts for the two sample periods are a reflection of the fact that 2011 was a much more volatile period than 2013. This is evidenced by the average volatility (as measured by the standard deviation

of daily returns) which amounts to 3.5% in the 2011 sample and 2.5% in the 2013 sample. The cross-sectional differences in volatility are substantial. The return standard deviation of the stock at the 95th percentile is more than three [more than five] times as large as the corresponding value for the stock at the 5th percentile in 2011 [2013].

The trading volume reported above relates to the total volume in the continuous trading session and the call auctions. On average the call auctions account for 12.2% of the total volume in the 2011 sample and for 11.9% in the 2013 sample. The closing auction is by far the most important auction in terms of volume. It accounts for 50.9% [58.7%] of the auction volume in the 2011 [2013] sample.⁶ The high volume share of the closing auction is consistent with the stylized fact that many institutional traders prefer to trade at or near the close (e.g. Cushing and Madhavan (2000)). The opening auction accounts for 25.5% [25.7%] of the auction volume, the intraday auction for 3.0% [3.9%] and auctions after volatility interruptions for the remaining 20.8% [11.7%]. When interpreting the last figure one needs to take into account that there are, on average, 0.51 [0.17] volatility interruptions per stock and day. Thus, conditional on a volatility interruption occurring, the call auction conducted to restart trading accounts for a considerable fraction of the volume.

The table also reveals the percentage of auctions with positive volume. Averaged over all stocks, 79.5% [82.1%] of the closing auctions, 71.3% [72.5%] of the opening auctions and 35.6% [55.1%] of the intraday auctions have non-zero volume in the 2011 [2013] sample. There is considerable cross-sectional variation. The stock at the 95th percentile always has positive volume in the opening and closing auction. In contrast, the stock at the 5th percentile has positive volume in the opening and closing auctions on less than one third of the trading days in the sample period. The fraction of intraday auctions with positive volume is even lower. These figures indicate that some of our sample stocks are very illiquid

⁶Note that these figures were obtained by first calculating the volume share for each stock and then averaging over all sample stocks. If we first aggregate the volume in the different auctions and then calculate the volume shares based on the aggregate volume, we obtain much higher volume shares for the closing auction. This is because the share of the closing auction is higher for stocks with higher total auction volume.

indeed.

While the number of opening, intraday and closing auctions is fixed at one per stock and day, the same does not hold for auctions after volatility interruptions. As noted above, these auctions are triggered by large price changes. Their number is thus endogenous. Therefore, the figures provided for this auction type have to be interpreted differently. They indicate that, on average, there were 0.51 [0.17] such auctions per stock and day in 2011 [2013]. The higher number of volatility auctions in the 2011 sample is a reflection of the higher volatility in 2011 already documented above. Again, there is considerable cross-sectional variation. The stock at the 5th percentile has 0.07 [0] auctions after volatility interruptions per day while the stock at the 95th percentile has 1.20 [0.64] such auctions per day in 2011 [2013]. These numbers reflect the large differences in return volatility across the sample stocks documented above.

Table 1 about here

Table 2 provides more insight into the relative importance of the four different types of auctions. It shows the fraction of the total trading volume accounted for by the call auctions for five groups of stocks. The groups are created as follows. Those stocks that are categorized as liquid by the exchange are sorted into one group denoted "Liquid". As noted in section 2 these stocks do not require a market maker to be traded continuously. The remaining stocks are sorted into size quartiles.

The table reveals a remarkable pattern. The volume share of the closing auction increases almost monotonically as we move from small to larger stocks. In 2011 [2013] the closing auction accounts for 3.16% [2.05%] of the total volume in stocks of the smallest size-quartile. This fraction increases to 7.4% [9.46%] for the stocks in the largest volume quartile and to 10.30% [15.40%] in the "liquid" group. A potential explanation for this finding is that (as will

⁷Note that our dataset only contains information on auctions after volatility interruptions with positive volume. It is conceivable that there are additional volatility interruptions in which the auction conducted to restart trading yielded zero volume. We do not observe these cases.

be discussed in the next section) principal traders have a much higher market share in large and high volume stocks and, at the same time, display a revealed preference for trading in the closing auction (possibly because they use the closing price as benchmark and/or want to avoid holding inventory overnight). Consequently, the closing auctions accounts for a higher fraction of the total volume in larger stocks. We find the opposite pattern for the opening and intraday auctions. Here, the auction accounts for a higher share of the total volume for smaller stocks than it does for larger ones. The same is true for auctions after volatility interruptions. For these auctions, the pattern is partially explained by the fact that smaller stocks tend to have more volatile returns and therefore experience more frequent volatility interruptions.⁸

Table 2 about here

5 Results

5.1 Market Shares and Trading Patterns

We start our empirical analysis by considering the participation of the three groups of traders (agency traders, principal traders, and designated market makers) in the call auctions. The results are shown in Table 3. Panel A shows results for groups of stocks sorted by market capitalization (using the procedure described above) while Panel B presents results for the four different types of call auctions. In each panel, columns 1-6 show the results for the 2011 sample while columns 7-12 show those for the 2013 sample. Columns 1-3 and 7-9 (denoted "any participation") show how often traders of the respective group participate in an auction. Take as an example the figure 89.54 in the upper left cell of Panel A. It indicates that agency

⁸In fact, the number of auctions after volatility interruptions decreases almost monotonically across the size groups.

⁹We use the term *participation rate* to measure how frequently the traders in a group participate in an auction. We use the term *participation share* to measure the fraction of the auction volume the traders in a group account for.

traders participated in 89.54% of the auctions for the group of stocks of the smallest firms. Columns 4-6 and 10-12 show market shares as a percentage of the Euro trading volume. The figure 69.84 in column 4 of the first line in Panel A indicates that transactions by agency traders account for 69.84% of the volume in the auctions for the stocks of the smallest firms.

Table 3 about here

Agency traders are the most active auction participants. They participate in more than 89% of the auctions even in the stocks of the smallest size group. Principal traders participate less in auctions for small-cap stocks. Their participation rates increase monotonically as we move towards more liquid stocks. The pattern for designated market makers is the exact opposite. In line with hypothesis 1, they have the highest participation rates (37.0% in the 2011 sample and 25.0% in the 2013 sample) in the smallest firms, whereas their participation rates are only 13.6% and 9.0% in the 2011 and 2013 sample, respectively, for the group of "liquid" stocks. Designated market makers thus appear to provide relatively more liquidity to stocks that are inherently illiquid.

When considering market shares in terms of the Euro trading volume we find that trading is dominated by agency and principal traders. Together they account for 90.2% to 99.7% of the trading volume. Considering the five groups of stocks reveals a remarkable pattern. Agency traders have the largest market shares in the least liquid market capitalization group. Their market share then decreases monotonically as we move to the more liquid sample stocks. For principal traders we observe the reverse pattern. As for the agency and principal traders, the market shares of the designated market makers vary systematically across size groups. They account for 9.8% (2011) and 4.4% (2013), respectively, of the volume in the smallest size group. Consistent with hypothesis 1, their market shares decrease almost monotonically across the size groups and amount to only 0.3% (0.1%) in the 2011 (2013) sample in the most liquid group.

The market shares for the different auction types displayed in Panel B of Table 3 show a distinct pattern. Agency traders dominate the opening auctions and the call auctions after volatility interruptions. Principal traders, by contrast, dominate the closing auctions. Designated market makers have a low market share in the closing auction. It amounts to 0.35% of the Euro volume in the 2011 sample and to 0.21% in the 2013 sample. Their share in the opening auction is much larger, at 2.1% [1.6%] in the 2011 [2013] sample. Their market shares are highest in the intraday auctions (7.1% and 1.8%) and in auctions after volatility interruptions (4.5%) and (4.5%). It is interesting to compare the results on market maker participation to those on the market shares of the four auction types shown in Table 1. Market maker participation appears to be inversely related to the total volume transacted in the auction. In particular, the closing auction is by far the most important auction in terms of volume, but it has the lowest market maker participation share. The intraday auction, in contrast, is the least important auction in terms of volume and has the highest market maker participation share in the 2011 sample and the second highest share in the 2013 sample. This pattern is consistent with hypothesis 2. It also appears that market makers have a comparably high participation share in those auctions where price uncertainty is likely to be higher, namely, opening auctions and auctions after volatility interruptions. This finding is in line with hypothesis 3. This may, of course, partly be due to the market makers' affirmative obligation to participate in call auctions. While agency and principal traders can withdraw from the markets in times of high uncertainty, market makers are bound to stay.

The analysis so far has focused on market maker participation across the four different auction types. A related question is whether, within one type of auction, market maker participation varies systematically with the trading volume in the auction. To address this issue we estimate panel regressions in which we regress the market maker participation share on the logarithm of the auction volume. We estimate separate regressions for each auction type and we include firm fixed effects. Therefore, identification comes from the

within-stock variation in the auction volume. The results are presented in Table 4. For three of the four auction types (opening and closing auctions and auctions after volatility interruptions) the coefficient on the log auction volume is significantly negative. Thus, designated market makers participate relatively more when the trading volume in the auction is lower. This result is consistent with hypothesis 2. For the intraday auctions we obtain a positive coefficient, implying that market makers trade more when the auction volume is higher. It should be kept in mind, though, that the volume of the intraday auctions is generally very low. As shown in Table 1, they account for less than 4% of the total call auction volume on average.

Table 4 about here

The fixed effects panel regressions do not allow us to identify the effect of variables with no or little time-series variation. We therefore also estimate cross-sectional regressions. Table 5 reports the results of a cross-sectional fractional logit model. The dependent variable is the average participation share of the designated market makers. The independent variables include the log of the average market value of equity during the sample period, the share turnover (defined as the ratio of share trading volume and market capitalization¹⁰), the 5-minute price impact (averaged over the sample period) as a measure of adverse selection risk, and three dummy variables which indicate whether the stock is a constituent stock of the mid cap index MDAX, the technology stock index TecDAX, or the small cap index SDAX. The coefficients are the marginal effects in percentage points, measured per standard deviation for the continuous independent variables. Table 5 reports separate results for the opening, the closing and the intraday auctions, and for auctions after volatility interruptions. Columns 1-4 [5-8] show the results for the 2011 [2013] sample.

Consistent with hypothesis 1, market maker participation rates are decreasing in the market capitalization and in the turnover ratios. A coefficient of -1.5 for market capitalization implies

 $^{^{10}}$ The correlation between market capitalization and turnover is 0.26 (-0.11) in the 2011 (2013) sample.

that a one standard deviation increase in the log of market capitalization decreases market maker participation by 1.5 percentage point. Relative to the participation rates shown in Table 3 (all well below 10%) this is an economically significant decrease. In the 2013 sample there is, in line with hypothesis 3, some evidence that market maker participation rates are higher for stocks with higher adverse selection risk as measured by the 5-minute price impacts. The respective coefficients in the 2011 sample are insignificant. The results further imply market maker participation tends to be lower in the component stocks of the MDAX and TECDAX indices (which are the largest and most liquid stocks in our sample).

Table 5 about here

The results presented thus far suggest that designated market maker participation is higher in those auctions where trading interest from agency and principal traders is lower and where price uncertainty is likely to be higher. We elaborate on this point by analyzing the time-series determinants of market maker participation using logit regressions. We define a dummy variable that is set to 1 if the designated market maker buys or sells shares in an auction and is 0 otherwise. We then regress this dummy variable on the absolute change in the quote midpoint in the 5-minute interval of the continuous trading session immediately preceeding the call auction. We repeat the analysis using 30-minute intervals instead of 5-minute intervals. Given the differences between the auction types documented above, we estimate separate models for closing auctions, intraday auctions, and auctions after volatility interruptions.¹¹

The results for our two samples, the 2011 and the 2013 sample, are qualitatively similar. Therefore we report the results for the pooled data set here. They are presented in Table

¹¹Note that opening auctions cannot be included in the analysis because we do not observe price changes prior to the opening auction. Note further that the number of observations differs between the three auction types. This is because, as documented in Table 1, the fraction of auctions with non-zero volume differs across auction types. For auctions after volatility interruptions the number of observations is lower when we include the price change in a 30-minute interval as explanatory variable because some volatility interruptions occur during the first 30 minutes of the trading day.

6.¹² The coefficients on the absolute price change in the 5-minute and 30-minute intervals prior to the auctions are always positive and are significant in five out of six cases. This indicates that, consistent with hypothesis 3, the designated market makers are more likely to participate in auctions that take place at times of elevated volatility. The magnitude of the marginal effects implies that a one standard deviation change in the absolute midquote return increases the likelihood of market maker participation by 0.5 to 1.2 percentage points. These numbers should be related to the average market maker participation rates (shown in Table 3) which range between 8.8% (for the intraday auctions in the 2013 sample) and 37.6% (for auctions after volatility interruptions in the 2013 sample). Thus, in relative terms the likelihood of market maker participation increases by 1.4% to 6.3% upon a one standard deviation change in the absolute midquote return.

Table 6 about here

5.2 Price Stabilization

One important aspect of market making is whether market makers stabilize or destabilize prices. The NYSE specialists had the explicit affirmative obligation to preserve price continuity (e.g. Hasbrouck and Sofianos (1993), Panayides (2007)). We therefore analyze whether designated market makers exacerbate or dampen price changes. We proceed as follows. We calculate the signed price change from the last quote midpoint prior to the auction to the auction price. We then regress this variable on the signed market maker participation, defined as the market maker's trade in the auction divided by the total volume of the auction. In an alternative model specification we also include the quote midpoint return in the 5 [30] minutes prior to the auctions.¹³ We estimate separate regressions for the intraday auctions,

¹²We also estimate Tobit models where the dependent variable is the fraction of the auction volume accounted for by trades of designated market makers. The results are consistent with those shown in Table 6 and are therefore omitted.

¹³Below we report results for pooled OLS estimation. We also estimate panel regressions with (a) firm fixed effects and (b) firm and quarter fixed effects. The results are very similar to those reported below and are thus omitted.

closing auctions, and for auctions after volatility interruptions. We cannot report results for the opening auction because the dependent variable (the price change from the previous quote midpoint to the auction price) is not defined for the opening auction.

The results are shown in Table 7. The coefficient on the signed market share of designated market makers is always negative and is significant at the 1% level in every case. This implies, in line with hypothesis 4, that designated market makers sell shares when prices go up and buy shares when prices go down. E.g., in the hypothetical case that they are the buyers of half of the trading volume in an auction after a volatility interruption, the regression results suggest that the immediate return will be about 30 basis points lower than in an otherwise similar auction without market maker participation. Thus, designated market makers appear to "lean against the wind" and stabilize prices.

Table 7 about here

The analysis documented in Table 7 considers short-term price stabilization. In order to broaden the perspective we now consider close-to-open (overnight) returns and open-to-close returns. We regress the close-to-open returns on the market makers' signed buying volume in the closing auction of day (t-1) and the opening auction on day t. A negative close-to-open return can obtain when there is negative fundamental news, or when there is transitory upward pressure on prices in the closing auction and/or transitory downward pressure on prices in the opening auction. In the latter cases, a price-stabilizing market maker would sell in the closing auction and/or buy in the opening auction. Consequently, we expect a positive coefficient on the market makers' net buying volume in the closing auction of day (t-1) and a negative coefficient on the net buying volume on the opening auction on day t. A similar argument (resulting in the same expected coefficient signs) applies in the case of positive overnight returns.

In a second regression we regress the open-to-close return on the market makers' signed net buying volume in the opening and closing auctions of day t. By a similar argument as above,

price stabilization implies a positive coefficient on the market makers' net buying volume in the opening auction and a negative coefficient on the signed net buying volume in the closing auction.

The results are shown in Table 8. They are fully consistent with price stabilization and thus provide additional support for hypothesis 4. Market makers, on average, trade "against the trend". When the overnight return is positive, they have been buying in the closing auction, and they sell in the opening auction (and vice versa for negative overnight returns). Thus, their trading activity in both auctions contributes to a reduction of the magnitude of the overnight return. We find similar results for open-to-close returns. When the return is positive, market makers have been buying in the opening auction and selling in the closing auction. The results presented in Table 8 thus corroborate those documented in Table 7. Designated market makers trade in call auctions in a way that contributes to price continuity.

Table 8 about here

5.3 Profitability

An obvious question is whether designated market makers earn trading profits. To address this question we calculate price impacts, measured by the return from the auction price to a post-auction benchmark price. We use three different benchmark prices. The first is the quote midpoint 5 minutes after the auction, the second is the quote midpoint one hour after the auction and the third is the closing price of the trading day. Obviously, none of these benchmark prices is defined for the closing auction. Therefore, we assess the profitability of trades in the closing auction by relating the closing price to the opening price on the next trading day. Price impacts after sales are multiplied by (-1). Therefore, positive [negative] price impacts always imply profitable [unprofitable] trades.

In Table 9 we report the overall value-weighted mean, standard deviation and median price impact (columns 10-12). More importantly, we also report price impacts for trades conducted

by agency traders, designated market makers, and principal traders in the auction (columns 1-3, 4-6 and 7-9, respectively). We report separate results for buys and for sales.¹⁴

Results for the opening auctions are reported in Panel A. There appears to be a negative price drift after the opening auction which continues over the day. On average prices drop by 10.9 basis points in the 5 minutes after the auction. The price decline increases to 16.8 basis points after one hour and to almost 42 basis points until the close of trading. Agency traders suffer substantial losses when they buy (62.4 basis points until the close of trading). They make small profits when they sell, but these profits are insufficient to compensate the losses they incur after buying. Principal traders, on the other hand, earn substantial profits when they sell. These profits overcompensate the small losses they incur when they buy. Designated market makers are the only group that manages to buy profitably. The 5-minute price impact after market maker buys is negative, but it turns positive after one hour. The 5-minute price impact after market maker sales is essentially zero. It turns negative (at 9.2 basis points) after one-hour, implying a loss for market makers. However, this loss is reversed by the end of the trading day. The price impact measured with respect to the closing auction is 39 basis points. We thus conclude that designated market makers trade profitably in the opening auctions.

Table 9 about here

Panel B shows the results for auctions after volatility interruptions. The average price change after the auction is positive. In the five-minute interval after the auction the price increaes by 21 basis points. On average prices continue to rise. The average one-hour price impact is 36.9 basis points, and the price increases by 89.6 basis points until the close of trading. The majority of volatility interruptions are triggered by price declines. The positive price impact indicates that prices tend to revert after the trading halt, and that this reversal continues

 $^{^{14}}$ The results in Table 9 do not differentiate between "liquid" stocks (i.e. those that do not require a designated market maker to be traded continuously) and other stocks. We repeat the analysis for both groups of stocks separately and obtain consistent results.

for the entire trading day. When we consider the three groups of traders separately, we find that agency traders tend to lose money. Although they buy profitably, these profits are overcompensated by the losses they incur after selling. Principal traders, on the other hand, tend to buy profitably. They also make small short-term profits when selling, but the profit turns to a loss of 47.6 basis points when the closing price is used as benchmark. However, this loss is overcompensated by the gain after principal buys (63.3 basis points when the closing price is used as benchmark). The group that trades most profitably in auctions after volatility interruptions are the designated market makers. Prices increase strongly after market makers buy (by 26.1 basis points within 5 minutes, by 50.9 basis points within an hour and by 118 basis points until the end of the trading day). When designated market makers sell they earn small short term profits. These profits turn to a small loss until the end of the trading day. However, the magnitude of the loss is small relative to the profits that market makers earn after buying (14.7 basis points as compared to 118 basis points).

Panel C reports the results for the closing auctions. As noted above, we use the opening price on the next trading day as reference price to calculate the price impact. There is no noticeable price drift in the overnight period. On average prices drop by less than one basis point. Agency traders appear to win. They earn 3.4 basis points when they buy and 4.2 basis points when they sell. This pattern is reversed for principal traders. Market makers suffer a small loss of 4.8 basis points when they buy in the closing auction. This loss, however, is overcompensated by the 24.4 basis points which they gain on average after selling in the closing auction.

Panel D shows the results for the regular intraday auctions. Price impacts are slightly lower than those after the opening auction. On average prices drop by 5 basis points. Agency traders lose slightly, principal traders roughly break even, and market makers trade profitably.

To put the market maker profits into perspective we compare them to the bid-ask spread in the continuous trading session. To this end, we add up the price impact after market maker buys and sales (using the closing price or, in the case of the closing auction, the next opening price as benchmark). We obtain mean [median] values of 11.5 [29.1] basis points for the opening auction, 19.6 [14.6] basis points for the closing auction, 82.9 [25.3] basis for the regular intraday auctions and 103.3 [82.5] basis points for auctions after volatility interruptions. The mean [median] bid-ask spread in the continuous trading session amounts to 109.8 [61.6] basis points in the 2011 sample and 73.5 [40.1] basis points in the 2013 sample (see Table 1). Thus, market maker profits in the call auctions appear to be of a reasonable magnitude. However, considering percentage numbers only does not allow us to estimate the profits that market makers actually earn. Therefore, Table 10 shows the average Euro profit per auction. The figures reveal that market makers, conditional on participating in a closing auction, earn 11.77 Euro on average. The corresponding figures for the intraday auctions, opening auctions and auctions after volatility interruptions are 25.46 Euros, 22.28 Euros and 31.84 Euros, respectively. These figures translate into earnings per stock and year of 2,623.84 Euros. The modest size of this number 15 supports the conclusion that market makers are not extracting excessive rents from their activity.

Table 10 about here

The results of the profitability analysis can be summarized as follows. Consistent with hypothesis 5 market makers trade profitably in the call auctions. Their profits exceed those of principal traders (while agency traders lose money on average). The profits of market makers are a reflection of the fact that they provide liquidity to other traders, and that liquidity supply is a valuable service that has its price. Some simple back-of-the-envelope calculations suggest that market maker profits are not excessive.

¹⁵Note that this is the estimated gross profit from the market makers' trading activity. The cost of doing business has not yet been deducted.

6 Conclusion

Many equity markets employ designated market makers to supply additional liquidity for small and mid cap stocks, and they use a hybrid trading system that combines a continuous trading session and call auctions. In this paper we use data from Germany's Xetra system to analyze designated market maker activity in the call auctions. There are four different types of call auctions in Xetra - opening auctions, closing auctions, one regular intraday auction per stock and per day, and auctions after volatility-induced trading halts. Taken together, the call auctions account for approximately 12% of the total trading volume of the stocks in our sample.

We find that trading in the call auctions is dominated by principal traders and agency traders while the share of designated market makers is much lower. We further find that their share in the total auction volume is inversely related to firm size. Thus, in the cross-section market makers tend to trade relatively more in less liquid stocks. A similar pattern holds across the four auction types. Market makers are relatively least active in the closing auction (which is by far the most liquid of the four auction types) and are relatively more active in the less liquid auctions. Taken together it appears that market makers trade relatively more actively precisely in those stocks and on those occasions where their contribution to liquidity is needed more. Results of a time-series analysis provide further support for this interpretation, revealing that market makers tend to trade more in times of higher volatility. These results are likely, at least in part, the result of their affirmative obligation to quote. When we relate returns to signed market maker trading activity we find clear evidence that market makers stabilize prices. They thus contribute to price continuity.

We further document that market maker trades are profitable. Prices increase on average after market makers buy, and they decrease on average after market makers sell. Taken together our results indicate that designated market makers provide a valuable service to the market. The trading profits which they earn can be interpreted as the implicit price that

they charge for this service.

References

- Abad, D. and R. Pascual (2010). Switching to a temporary call auction in times of high uncertainty. *Journal of Financial Research* 33, 45–75.
- Amihud, Y., H. Mendelson, and B. Lauterbach (1997). Market microstructure and securities values: Evidence from the tel aviv stock exchange. *Journal of Financial Economics* 45, 365–390.
- Anand, A., C. Tanggaard, and D. Weaver (2009). Paying for market quality. *Journal of Financial and Quantitative Analysis* 44 (6), 1427–1457.
- Anand, A. and K. Venkataraman (2016). Market conditions, fragility, and the economics of market making. *Journal of Financial Economics* 121(2), 327–349.
- Bessembinder, H., J. Hao, and K. Zheng (2015). Market making contracts, firm value, and the ipo decision. *Journal of Finance* 70(5), 1997–2028.
- Chang, R., G. Rhee, G. Stone, and N. Tang (2008). How does the call market method affect price efficiency? evidence from the singapore stock market. *Journal of Banking and Finance* 32, 2205–2219.
- Chelley-Steeley, P. (2008). Market quality changes in the london stock market. *Journal of Banking and Finance* 32, 2248–2253.
- Chelley-Steeley, P. (2009). Price synchronicity: The closing call auction and the london stock market. Journal of International Financial Markets, Institutions & Money 19, 777–791.
- Clark-Joseph, A. D., M. Ye, and C. Zi (2016). Designated market makers still matter: Evidence from two natural experiments. *Journal of Financial Economics*, forthcoming.
- Cohen, K. and R. Schwartz (1989). An electronic call market: Its design and desirability. In
 H. Lucas and R. Schwartz (Eds.), The challenge of information technology for the securities
 markets, Chapter 2, pp. 15–58. Homewood, Illinois: Irwin Professional Pub.

- Cushing, D. and A. Madhavan (2000). Stock returns and trading at the close. *Journal of Financial Markets* 3, 45–67.
- Declerck, F. and P. Hazart (2002). Impacts de l'animation sur la qualité du second marché.

 Banque et Marché 60, 5–18.
- Economides, N. and R. Schwartz (1995). Electronic call market trading. *Journal of Portfolio Management 21*, 10–18.
- Eldor, R., S. Hauser, B. Pilo, and I. Shurki (2006). The contribution of market makers to liquidity and efficiency of options trading in electronic markets. *Journal of Banking & Finance* 30(7), 2025–2040.
- Hasbrouck, J. and G. Sofianos (1993). The trades of market makers: An empirical analysis of nyse specialists. *Journal of Finance* 48(5), 1565–1593.
- Hengelbrock, J. (2011). Designated sponsors and bid-ask spreads on Xetra. Available at SSRN 1046961.
- Kalay, A., L. Wei, and A. Wohl (2002). Continuous trading or call auctions: Revealed preference of investors at the tel aviv stock exchange. *Journal of Finance* 57, 523–542.
- Kandel, E., B. Rindi, and L. Bosetti (2012). The effect of a closing call auction on market quality and trading strategies. *Journal of Financial Intermediation* 21, 23–49.
- Kehr, C., J. Krahnen, and E. Theissen (2001). The anatomy of a call market. *Journal of Financial Intermediation* 10(3), 249–270.
- Madhavan, A. and V. Panchapagesan (2000). Price discovery in auction markets: A look inside the black box. *Review of Financial Studies* 13, 627–658.
- Menkveld, A. J. and T. Wang (2013). How do designated market makers create value for small-caps? *Journal of Financial Markets* 16(3), 571–603.

- Nimalendran, M. and G. Petrella (2003). Do "¿œthinly-traded"; œstocks benefit from specialist intervention? *Journal of banking & finance* 27(9), 1823–1854.
- Pagano, M., L. Peng, and R. Schwartz (2013). A call auction's impact on price formation and order routing: Evidence from the nasdaq stock market. *Journal of Financial Markets*, 331–361.
- Pagano, M. and R. Schwartz (2003). A closing call's impact on market quality at euronext paris. *Journal of Financial Economics* 68, 439–484.
- Panayides, M. (2007). Affirmative obligations and market making with inventory. *Journal* of Financial Economics 86(2), 513–542.
- Sabourin, D. (2006). Are designated market makers necessary in centralized limit order markets? Available at SSRN 889774.
- Schwartz, R. (2000). Building a better stock market: New solutions to old problems. *Available at SSRN 243536*.
- Skjeltorp, J. A. and B. A. Ødegaard (2015). When do listed firms pay for market making in their own stock? *Financial Management* 44(2), 241–266.
- Theissen, E. and C. Westheide (2015). Designated market maker ratings. Working Paper.
- Venkataraman, K. and A. Waisburd (2007). The value of the designated market maker.

 Journal of Financial and Quantitative Analysis 42(3), 735.
- Zimmermann, K. (2014). Price discovery in european volatility interruprions. Available at SSRN 2365772.

Table 1: Descriptive statistics

| | | | 2011 | | | | | 2013 | | |
|------------------------------|---------|---------|-----------|-------------|-------------------------------|---------|---------|-----------|---------|----------|
| | Mean | Median | Std. dev. | 5th pct | 95th pct | Mean | Median | Std. dev. | 5th pct | 95th pct |
| | | | Compan | y and stoc | Company and stock statistics | | | | | |
| Market cap. (mio euros) | 689.801 | 219.308 | 1062.131 | | 3050.164 | 993.122 | 293.850 | 1565.321 | 30.328 | 4715.898 |
| Price (euros) | 21.681 | 12.573 | 29.486 | | 69.876 | 25.770 | 15.481 | 33.976 | 1.252 | 82.154 |
| Volatility (daily std) | 0.035 | 0.033 | 0.013 | 0.016 | 0.054 | 0.025 | 0.019 | 0.021 | 0.011 | 0.058 |
| Price impact 5m | 0.539 | 0.395 | 0.473 | | 1.415 | 0.398 | 0.228 | 0.506 | 0.052 | 1.184 |
| Bid-ask spread | 1.098 | 0.616 | 1.088 | | 3.164 | 0.735 | 0.401 | 0.810 | 0.070 | 2.488 |
| ADV total (euros) | 2796464 | 372706 | 5972757 | 11719 | 14666964 | 2151163 | 315215 | 4280445 | 17166 | 9601774 |
| | | | Call an | ction trade | Call auction trade statistics | | | | | |
| Volume share of auctions | 12.23 | 10.68 | 5.81 | 90.9 | 24.83 | 11.93 | 10.62 | 5.22 | 5.27 | 22.16 |
| ADV Open (euros) | 37181 | 9275 | 74923 | 443 | 172748 | 30523 | 8829 | 58957 | 455 | 135793 |
| ADV Close (euros) | 271666 | 15356 | 625619 | 379 | 1788802 | 303031 | 16547 | 676227 | 504 | 1529346 |
| ADV Intraday (euros) | 6950 | 1221 | 12363 | 24 | 36761 | 3976 | 1472 | 2692 | 21 | 16610 |
| ADV Volatility (euros) | 8170 | 4500 | 13292 | 444 | 25838 | 4601 | 1723 | 10885 | 100 | 18225 |
| Open share of auctions | 25.45 | 23.80 | 15.07 | 5.31 | 52.59 | 25.74 | 22.70 | 17.07 | 4.50 | 58.50 |
| Close share of auctions | 50.95 | 49.40 | 26.24 | 11.85 | 92.13 | 58.74 | 60.31 | 27.01 | 14.66 | 94.26 |
| Intraday share of auctions | 3.01 | 2.29 | 2.71 | 0.41 | 9.46 | 3.88 | 2.56 | 4.13 | 0.50 | 10.91 |
| Volatility share of auctions | 20.85 | 14.02 | 20.11 | 0.20 | 61.87 | 11.69 | 4.15 | 15.97 | 0.00 | 47.58 |
| Intraday > 0 | 35.62 | 27.27 | 31.90 | 0.00 | 93.18 | 55.07 | 62.22 | 36.19 | 2.22 | 100.00 |
| Close > 0 | 79.54 | 97.73 | 27.75 | 27.27 | 100.00 | 82.13 | 100.00 | 26.06 | 28.89 | 100.00 |
| Open > 0 | 71.26 | 79.55 | 28.14 | 22.73 | 100.00 | 72.45 | 82.22 | 28.45 | 22.22 | 100.00 |
| Volatility > 0 | 51.22 | 40.91 | 46.29 | 6.82 | 120.46 | 17.08 | 29.9 | 27.34 | 0.00 | 64.44 |

the average price impact during continuous trading, Bid-ask spread is the time-weighted average quoted spread in percent, and ADVcall auctions. Volume share of auctions is the percentage of call auction trades relative to total trading volume. Auction type share of auction volume is the relative share, in percent, of the respective auction type in total call auction trading. Auction type > 0 is is the market value of equity, Price is the stock price, and Volatility is the standard deviation of daily returns. Price impact 5m is total is the daily trading volume in euros aggregated over the continuous and call auction phases. The bottom panel of the table summarizes trading activity in call auctions. ADV auction type is the average daily Euro trading volume in the different types of the percentage of auction of the respective type with a non-zero trading volume. For volatility auctions, it is the number of volatility This table shows descriptive statistics for the firms in our sample. The top panel provides information on the companies, their stocks, and their trading overall. The left part of the table is summarizing data for our sample in 2011, the right part for 2013. Market cap. auction with trading volume expressed in percent.

Table 2: Market shares of types of call auctions by market capitalization quartile and liquidity class

| | | | 2011 | | |
|--------|------|-------|----------|------|--------------|
| | Open | Close | Intraday | Vola | All Auctions |
| Small | 3.67 | 3.16 | 0.38 | 5.41 | 12.63 |
| 2 | 3.78 | 2.89 | 0.22 | 2.02 | 8.91 |
| 3 | 2.70 | 4.60 | 0.31 | 1.46 | 9.06 |
| Big | 1.80 | 7.40 | 0.30 | 0.75 | 10.26 |
| Liquid | 1.18 | 10.30 | 0.22 | 0.15 | 11.84 |
| | | | 2013 | | |
| Small | 2.81 | 2.05 | 0.27 | 1.50 | 6.64 |
| 2 | 2.79 | 3.89 | 0.52 | 1.50 | 8.71 |
| 3 | 2.72 | 5.35 | 0.32 | 0.77 | 9.17 |
| Big | 1.59 | 9.46 | 0.29 | 0.35 | 11.69 |
| Liquid | 1.29 | 15.40 | 0.16 | 0.09 | 16.94 |

This table shows the fraction of trading, measured by the volume in euros, conducted in call auctions and relative to the total trading volume, for non-liquid stocks, sorted by market capitalization quartiles at the beginning of the two sample periods, and liquid ones.

Table 3: Market shares in call auctions

| | | | 2011 | | | | | | 20 | 2013 | | |
|------------|--------|-------------------|--|------------|--|-----------|------------|-------------------|---------|-------|-----------------|-------|
| | any | any participation | ation | vol | volume in euros | Iros | any | any participation | tion | lov | volume in euros | ros |
| | Agency | DMM | Principal | A | DMM | Ь | A | DMM | Ь | A | DMM | Ь |
| | | A: I | A: Participation by market capitalization quartile and liquidity class | by mark | et capital | ization q | uartile an | d liquidit | y class | | | |
| Small | 89.54 | 36.98 | 47.47 | 69.84 | 9.80 | 20.36 | 89.54 | 25.02 | 65.36 | 72.09 | 4.44 | 23.46 |
| 2 | 89.16 | 32.31 | 63.61 | 65.16 | 6.12 | 28.72 | 89.86 | 27.64 | 70.21 | 57.80 | 6.19 | 36.01 |
| 3 | 92.25 | 30.65 | 76.05 | 57.21 | 5.37 | 37.42 | 93.92 | 20.73 | 81.66 | 49.73 | 4.27 | 46.00 |
| Big | 94.13 | 27.74 | 85.07 | 44.91 | 3.03 | 52.06 | 94.81 | 15.30 | 91.34 | 43.61 | 1.69 | 54.71 |
| Liquid | 97.50 | 13.57 | 95.88 | 39.47 | 0.34 | 60.19 | 96.82 | 8.98 | 99.41 | 43.44 | 0.11 | 56.45 |
| | | | B | : Particip | B: Participation by type of call auction | type of c | all auctio | n | | | | |
| Open | 97.35 | 30.32 | 76.29 | 61.70 | 2.09 | 36.21 | 94.55 | 22.36 | 84.63 | 56.48 | 1.58 | 41.94 |
| Close | 91.03 | 22.13 | 95.19 | 37.10 | 0.35 | 62.54 | 93.58 | 15.26 | 96.74 | 42.14 | 0.21 | 57.65 |
| Intraday | 92.21 | 13.47 | 73.35 | 40.83 | 7.08 | 52.09 | 93.06 | 8.77 | 77.72 | 54.44 | 1.82 | 43.75 |
| Volatility | 92.02 | 36.07 | 56.78 | 67.14 | 4.49 | 28.37 | 94.50 | 37.61 | 60.05 | 64.90 | 4.04 | 31.06 |

capitalization quartiles at the beginning of the two sample periods, and liquid ones. Panel B reports results sorted by type of volume in Euros, for agency traders, designated market makers (DMM), and principal traders. The left part of the tables This table shows the fractions of call auctions with non-zero participation and the share of trading, measured by the trading shows data from 2011, the right part from 2013. Panel A reports results separately for non-liquid stocks, sorted by market call auction.

Table 4: Panel regression of participation share of designated market makers

| | Open | Close | Intraday | Volatility |
|--------------------|-----------|-----------|----------|------------|
| Log Auction Volume | -0.863*** | -1.362*** | 0.433*** | -3.755*** |
| | (-6.29) | (-10.15) | (2.61) | (-16.97) |
| Constant | 18.025*** | 18.836*** | 4.471*** | 44.530*** |
| | (19.91) | (18.32) | (5.07) | (33.87) |
| Observations | 15515 | 17445 | 9653 | 7785 |

This table shows the results of panel regressions, separately for each type of call auction, explaining the share of DMM trading. The regressions include stock fixed effects and the coefficients are expressed in percentage points. Standard errors are clustered at the stock level. ***, ** and * denote statistical significance at the 1%, 5% or 10% level.

Table 5: Cross-sectional fractional logit regression of participation shares of designated market makers

| | | 2011 | 11 | | | 2013 | 13 | |
|---------------------------|-----------|-----------|---------------|------------|-----------|-----------|-----------|----------------|
| | Open | Close | Intraday | Volatility | Open | Close | Intraday | Volatility |
| Log Market cap. | -1.542** | -1.296** | -3.671** | -1.369 | -1.559** | -1.836*** | -1.296 | 1.074 |
| | (-1.97) | (-2.11) | (-2.11) | (-1.45) | (-2.25) | (-4.41) | (-1.48) | (0.81) |
| Share turnover (relative) | -5.743*** | -2.849*** | -2.540^{**} | -3.423*** | -4.990*** | -2.583*** | -5.836*** | -3.614^{***} |
| | (-5.48) | (-3.60) | (-2.40) | (-3.19) | (-5.34) | (-4.51) | (-4.69) | (-2.67) |
| Price impact 5m | -0.839 | -0.056 | -1.947 | -0.072 | 0.564*** | 0.157 | ***909.0 | 0.614 |
| | (-1.04) | (-0.10) | (-0.97) | (-0.08) | (3.62) | (1.39) | (2.91) | (1.06) |
| MDAX | -3.396** | -3.514*** | 9.889*** | -5.592*** | -2.839** | -2.251** | -1.432 | -7.810*** |
| | (-2.57) | (-3.69) | (2.65) | (-3.80) | (-2.07) | (-2.39) | (-0.81) | (-3.13) |
| TecDAX | -0.826 | -2.998*** | -3.598*** | -3.670* | 0.203 | -1.131 | 1.041 | -4.575* |
| | (-0.49) | (-3.17) | (-3.01) | (-1.78) | (0.13) | (-1.20) | (0.49) | (-1.66) |
| SDAX | 1.907 | -0.393 | 3.191 | -0.737 | 2.606** | 0.419 | 0.189 | -3.388 |
| | (1.50) | (-0.38) | (1.49) | (-0.55) | (1.97) | (0.51) | (0.16) | (-1.38) |
| Observations | 250 | 250 | 229 | 246 | 209 | 209 | 207 | 174 |

maker trading in each type of call auction. The left part of the table provides results for 2011, the right part for 2013. The explanatory variables comprise the natural logarithm of the market value of equity as of the beginning of the respective sample period, share turnover, the ratio of total trading volume and the market value of equity, the average five minute MDAX, the technology stock index TexDAX, or the smallcap index SDAX. The independent variables, except for the This table shows marginal effects obtained from fractional logit regressions explaining the share of designated market price impact, and dummy variables indicating whether a stock is among the constituent stocks of the midcap index indicator variables, are standardized and the coefficients are expressed in percentage points. Standard errors are robust to heteroskedasticity. ***, ** and * denote statistical significance at the 1%, 5% or 10% level.

Table 6: Logit regressions of participation of designated market makers

| | Close | Close | Intraday | Intraday | Volatility | Volatility |
|-----------------|-------------------|--------------------|-------------------|--------------------|------------------|-------------------|
| Abs. return 5m | 0.517** (2.46) | | 0.539** (2.22) | | 0.996* (1.92) | |
| Abs. return 30m | | 1.238*** (4.52) | | 0.551^* (1.90) | | $0.530 \\ (0.65)$ |
| Observations | 17035 | 17035 | 9464 | 9462 | 7217 | 6408 |

This table shows marginal effects obtained from logit regressions explaining the participation of designated market makers in closing auctions, regular intraday auctions, and auctions after volatility-induced trading halts. The independent variable is the standardized absolute value of the quote midpoint return in the 5 or 30 minute interval immediately preceding the auction. Standard errors are clustered at the stock level. ***, ** and * denote statistical significance at the 1%, 5% or 10% level.

Table 7: Determinants of auction returns

| | Close | Close | Close | Intraday | 1 | Intraday Intraday | Volatility | Volatility | Volatility |
|---------------------|-----------------|----------------------|------------------------|----------------|-------------------------------|---|----------------|-------------------------------|----------------------------|
| DMM share (signed) | -0.332*** -0.33 | -0.337*** | -0.335*** | -0.245*** | -0.247*** | -0.247*** | -0.585*** | -0.603*** | -0.614*** |
| Return previous 5m | (-4.10) | -4 -0.18 -4 | (-4.14) | (-0.11) | (-0.20) -0.093* (-1.68) | (-0.11) | (-0.10) | -3.08) -0.090*** -3.08) | (50.6-) |
| Return previous 30m | | | -0.025 | | | -0.037** | | | -0.050** (20.08) |
| Constant | 0.031** (2.58) | 0.034^{***} (2.80) | 0.032^{***} (2.61) | 0.010 (1.38) | 0.011 (1.47) | $\begin{pmatrix} -290 \\ 0.012 \\ (1.63) \end{pmatrix}$ | -0.059 (-1.10) | -0.059 (-1.10) | (-2.36) -0.069 (-1.38) |
| Observations | 17035 | 17035 | 17035 | 9464 | 9464 | 9462 | 7456 | 7217 | 6408 |

maker trading volume divided by the total volume traded in the respective auction, and standardized quote midpoint returns This table shows results of OLS regressions explaining the signed return from the quote midpoint immediately prior to the auction to the auction price. Explanatory variables are the signed market maker participation, computed as the signed market in either the 5-minute or the 30-minute interval immediately preceeding the auction. Standard errors are clustered at the stock level. ***, ** and * denote statistical significance at the 1%, 5% or 10% level.

Table 8: Market Maker impact on Close-Open and Open-Close Returns

| | Close-Open | Open-Close |
|----------------------|------------|------------|
| DMM Open Net-Buying | -0.724*** | 0.327*** |
| | (-7.31) | (2.84) |
| DMM Close Net-Buying | 0.315*** | -0.506*** |
| | (3.39) | (-3.58) |
| Constant | -0.007 | 0.007 |
| | (-0.33) | (0.24) |
| Observations | 12159 | 12766 |

This table shows results of OLS regressions of the returns from the previous close to the open and from the open to the close as explained by designated market makers' signed trading, as a share of total volume in the respective opening and closing auction. Returns are adjusted for the corresponding sample average on the day. Standard errors are clustered at the stock level. ***, ** and * denote statistical significance at the 1%, 5% or 10% level.

Table 9: Price impacts when different participants trade in call auctions

| | | Agency | | | DMM | | | Principal | | | All | |
|----------|-------------------------|-----------|--------|--------|-----------|------------------------|---------|-----------|--------|--------|-----------|--------|
| | Mean | Std. dev. | Median | Mean | Std. dev. | Median | Mean | Std. dev. | Median | Mean | Std. dev. | Median |
| | | | | | A: 0 | A: Opening Auctions | ctions | | | | | |
| | | | | | | Buying | | | | | | |
| PI 5m | -0.155 | 1.323 | -0.083 | -0.046 | 1.223 | 0.024 | -0.035 | 1.071 | -0.008 | -0.109 | 1.235 | -0.040 |
| PI 1h | -0.253 | 2.410 | -0.039 | 0.100 | 1.858 | 0.071 | -0.039 | 2.033 | 0.006 | -0.168 | 2.272 | -0.023 |
| PI Cl. | -0.624 | 4.437 | -0.085 | 0.161 | 2.997 | 0.000 | -0.099 | 3.403 | 0.035 | -0.419 | 4.077 | 0.000 |
| | | | | | | Selling | | | | | | |
| PI 5m | 0.072 | 1.331 | 0.010 | 0.001 | 1.088 | 0.034 | 0.164 | 1.094 | 0.099 | 0.109 | 1.235 | 0.040 |
| PI 1h | 0.120 | 2.460 | -0.061 | -0.092 | 1.823 | 0.019 | 0.246 | 2.002 | 0.071 | 0.168 | 2.272 | 0.023 |
| PI Cl. | 0.233 | 4.221 | -0.045 | 0.390 | 3.095 | 0.291 | 0.681 | 3.900 | 0.161 | 0.419 | 4.077 | 0.000 |
| | | | | | B: Vc | B: Volatility Auctions | uctions | | | | | |
| | | | | | | Buying | | | | | | |
| PI 5m | 0.194 | 1.877 | 0.196 | 0.261 | 1.739 | 0.198 | 0.249 | 1.719 | 0.197 | 0.214 | 1.824 | 0.196 |
| PI 1h | 0.492 | 3.035 | 0.465 | 0.509 | 2.871 | 0.406 | 0.271 | 2.893 | 0.369 | 0.426 | 2.986 | 0.426 |
| PI Cl. | 0.995 | 4.655 | 0.848 | 1.180 | 4.456 | 0.825 | 0.633 | 4.897 | 0.589 | 0.896 | 4.723 | 0.758 |
| | | | | | | Selling | | | | | | |
| PI 5m | -0.364 | 1.918 | -0.313 | 0.143 | 1.801 | 0.184 | 0.096 | 1.533 | 0.010 | -0.214 | 1.824 | -0.196 |
| PI 1h | -0.666 | 3.049 | -0.674 | 0.090 | 2.868 | 0.131 | 0.071 | 2.776 | -0.046 | -0.426 | 2.986 | -0.426 |
| PI Cl. | -1.113 | 4.826 | -1.139 | -0.147 | 4.099 | 0.000 | -0.476 | 4.508 | -0.260 | -0.896 | 4.723 | -0.758 |
| (conting | (continued on next nage | ext nage) | | | | | | | | | | |

(continued on next page)

Table 9: Price impacts when different participants trade in call auctions (continued from previous page)

| | 1 | | 1 | | DMM | , | 1 | Principal | , | 1 | All | , |
|--------------|--------|-----------|--------|--------|-----------|----------------------|--------|-----------|--------|--------|-----------|--------|
| | Mean | Std. dev. | Median | Mean | Std. dev. | Median Mean | Mean | Std. dev. | Median | Mean | Std. dev. | Median |
| | | | | | C: C | C: Closing Auctions | ctions | | | | | |
| | | | | | | Buying | | | | | | |
| PI Op. 0.034 | 0.034 | 1.839 | 0.019 | -0.048 | 2.276 | 0.106 | -0.036 | 1.978 | 0.000 | -0.009 | 1.926 | 0.012 |
| | | | | | | Selling | | | | | | |
| PI Op. | 0.042 | 2.091 | -0.012 | 0.244 | 2.494 | 0.040 | -0.014 | 1.807 | -0.012 | 0.009 | 1.926 | -0.012 |
| | | | | | D: In | D: Intraday Auctions | ctions | | | | | |
| | | | | | | Buying | | | | | | |
| PI 5m | -0.081 | 0.472 | -0.068 | 0.178 | 0.484 | 0.151 | -0.041 | 0.369 | -0.060 | -0.053 | 0.420 | -0.059 |
| PI 1h | -0.025 | 0.943 | -0.044 | 0.498 | 1.160 | 0.232 | 0.261 | 0.805 | 0.067 | 0.145 | 0.887 | 0.020 |
| PI Cl. | -0.006 | 2.152 | 0.043 | 0.761 | 1.750 | 0.614 | 0.419 | 1.639 | 0.299 | 0.246 | 1.890 | 0.166 |
| | | | | | | Selling | | | | | | |
| PI 5m | 0.024 | 0.450 | 0.020 | 0.221 | 0.305 | 0.219 | 0.050 | 0.398 | 0.050 | 0.053 | 0.420 | 0.059 |
| PI 1h | -0.098 | | -0.030 | 0.109 | 0.583 | 0.139 | -0.243 | 0.873 | -0.063 | -0.145 | 0.887 | -0.020 |
| PI Cl. | -0.127 | 1.990 | -0.129 | 0.068 | 1.297 | 0.034 | -0.434 | 1.861 | -0.316 | -0.246 | 1.890 | -0.166 |

This table provides (separately for purchases and sales) information on returns (i.e. price impacts) earned by traders of the three different categories (agency traders, principal traders and designated market makers) when trading in the different types of call auctions. For closing auctions, returns are computed from the auction price to the next day's opening price. For the other auctions, returns are computed based on quote midpoints 5 minutes and 60 minutes after the auction, and for the return based on the same day's closing price.

Table 10: Market Maker Profit per auction

| | Close | Intraday | Open | Volatility |
|---------------|--------|----------|--------|------------|
| Profit (Euro) | 11.772 | 25.457 | 22.277 | 31.840 |
| Observations | 2238 | 919 | 3393 | 1966 |

This table shows the profits per auction earned by designated market makers in call auctions. The profit is calculated by multiplying the signed trade size by the return from the call auction to the closing price, with the exception of call auctions, for which the return is calculated to the next day's opening price.

crm working paper series



cfr working papers are available for download from www.cfr-cologne.de.

Hardcopies can be ordered from: centre for rinancial mesearch (сгм), albertus magnus platz, 50923 коеln, germany.

2017

| No. | Author(s) | Title |
|-------|---|--|
| 17-03 | G. Cici, S. Gibson, R. Moussawi | Explaining and Benchmarking Corporate Bond Returns |
| 17-02 | S. Jaspersen, P. Limbach | Knowing Me, Knowing You? Similarity to the CEO and Fund Managers' Investment Decisions |
| 17-01 | J.Grammig, EM. Küchlin | A two-step indirect inference approach to estimate the long-run risk asset pricing model |
| 2016 | | |
| No. | Author(s) | Title |
| 16-12 | A.Betzer, M. Ibel, H.S. Lee, P. Limbach, J.M. Salas | Are Generalists Beneficial to Corporate Shareholders? Evidence from Sudden Deaths |
| 16-11 | P. Limbach, M. Schmid, M. Scholz-Daneshgari | Do CEOs Matter? Corporate Performance and the CEO Life Cycle |
| 16-10 | V. Agarwal, R. Vashishtha, M. Venkatachalam | Mutual fund transparency and corporate myopia |
| 16-09 | MA. Göricke | Do Generalists Profit from the Fund Families' Specialists? Evidence from Mutual Fund Families Offering Sector Funds |
| 16-08 | S. Kanne, O. Korn, M.Uhrig-Homburg | Stock Illiquidity, Option Prices and Option Returns |
| 16-07 | S. Jaspersen | Market Power in the Portfolio: Product Market Competition and Mutual Fund Performance |
| 16-06 | O. Korn, MO. Rieger | Hedging With Regret |
| 16-05 | E. Theissen, C. Westheide | Call of Duty: Designated Market Maker Participation in Call Auctions |
| 16-04 | P. Gomber, S. Sagade, E. Theissen, M.C. Weber, C. Westheide | Spoilt for Choice: Order Routing Decisions in Fragmented Equity Markets |
| 16-03 | T.Martin, F. Sonnenburg | Managerial Ownership Changes and Mutual Fund Performance |
| 16-02 | A.Gargano, A. G. Rossi, R. Wermers | The Freedom of Information Act and the Race Towards Information Acquisition |
| 16-01 | G. Cici, S. Gibson, C. | Cross-Company Effects of Common Ownership: Dealings |

| | Rosenfeld | Between Borrowers and Lenders With a Common Blockholder |
|-------|---|--|
| 2015 | | |
| No. | Author(s) | Title |
| 15-17 | O. Korn, L. Kuntz | Low-beta Strategies |
| 15-16 | D. Blake, A.G. Rossi, A. Timmermann, I. Tonks, R. Wermers | Network Centrality and Pension Fund Performance |
| 15-15 | S. Jank, E. Smajbegovic | Dissecting Short-Sale Performance: Evidence from Large Position Disclosures |
| 15-14 | M. Doumet, P. Limbach, E. Theissen | Ich bin dann mal weg: Werteffekte von Delistings deutscher Aktiengesellschaften nach dem Frosta-Urteil |
| 15-13 | G. Borisova, P.K. Yadav | Government Ownership, Informed Trading and Private Information |
| 15-12 | V. Agarwal, G.O. Aragon, Z. Shi | Funding Liquidity Risk of Funds of Hedge Funds: Evidence from their Holdings |
| 15-11 | L. Ederington, W. Guan, P.K. Yadav | Dealer spreads in the corporate Bond Market: Agent vs. Market-Making Roles |
| 15-10 | J.R. Black, D. Stock, P.K. Yadav | The Pricing of Different Dimensions of Liquidity: Evidence from Government Guaranteed Bank Bonds |
| 15-09 | V. Agarwal, H. Zhao | Interfund lending in mutual fund families: Role of internal capital markets |
| 15-08 | V. Agarwal, T. C. Green, H. Ren | Alpha or Beta in the Eye of the Beholder: What drives Hedge Fund Flows? |
| 15-07 | V. Agarwal, S. Ruenzi, F. Weigert | Tail risk in hedge funds: A unique view from portfolio holdings |
| 15-06 | C. Lan, F. Moneta, R. Wermers | Mutual Fund Investment Horizon and Performance |
| 15-05 | L.K. Dahm, C. Sorhage | Milk or Wine: Mutual Funds' (Dis)economies of Life |
| 15-04 | A. Kempf, D. Mayston, M. Gehde-Trapp, P. K. Yadav | Resiliency: A Dynamic View of Liquidity |
| 15-03 | V. Agarwal, Y. E. Arisoy, N. Y. Naik | Volatility of Aggregate Volatility and Hedge Funds Returns |
| 15-02 | G. Cici, S. Jaspersen, A. Kempf | Speed of Information Diffusion within Fund Families |
| 15-01 | M. Baltzer, S. Jank, E. Smajlbegovic | Who trades on momentum? |
| 2014 | | |
| No. | Author(s) | Title |
| 14-15 | M. Baltzer, S. Jank, E.Smajlbegovic | Who Trades on Monumentum? |
| 14-14 | G. Cici, L. K. Dahm, A. Kempf | Trading Efficiency of Fund Families: Impact on Fund Performance and Investment Behavior |

| 14-13 | V. Agarwal, Y. Lu, S. Ray | Under one roof: A study of simultaneously managed hedge funds and funds of hedge funds |
|---|---|---|
| 14-12 | P. Limbach, F. Sonnenburg | Does CEO Fitness Matter? |
| 14-11 | G. Cici, M. Gehde-Trapp, M. Göricke, A. Kempf | What They Did in their Previous Life: The Investment Value of Mutual Fund Managers' Experience outside the Financial Sector |
| 14-10 | O. Korn, P. Krischak, E. Theissen | Illiquidity Transmission from Spot to Futures Markets |
| 14-09 | E. Theissen, L. S. Zehnder | Estimation of Trading Costs: Trade Indicator Models Revisited |
| 14-08 | C. Fink, E. Theissen | Dividend Taxation and DAX Futures Prices |
| 14-07 | F. Brinkmann, O. Korn | Risk-adjusted Option-implied Moments |
| 14-06 | J. Grammig, J. Sönksen | Consumption-Based Asset Pricing with Rare Disaster Risk |
| 14-05 | J. Grammig, E. Schaub | Give me strong moments and time – Combining GMM and SMM to estimate long-run risk asset pricing |
| 14-04 | C. Sorhage | Outsourcing of Mutual Funds' Non-core Competencies |
| 14-03 | D. Hess, P. Immenkötter | How Much Is Too Much? Debt Capacity And Financial Flexibility |
| | | |
| 14-02 | C. Andres, M. Doumet, E. Fernau, E. Theissen | The Lintner model revisited: Dividends versus total payouts |
| 14-02 14-01 | Fernau, E. Theissen | The Lintner model revisited: Dividends versus total payouts Corporate Governance and the Nature of Takeover Resistance |
| | Fernau, E. Theissen N.F. Carline, S. C. Linn, P. | |
| 14-01 | Fernau, E. Theissen N.F. Carline, S. C. Linn, P. | |
| 14-01 | Fernau, E. Theissen N.F. Carline, S. C. Linn, P. K. Yadav | Corporate Governance and the Nature of Takeover Resistance |
| 14-01 2013 No. | Fernau, E. Theissen N.F. Carline, S. C. Linn, P. K. Yadav Author(s) R. Baule, O. Korn, S. | Corporate Governance and the Nature of Takeover Resistance Title Which Beta is Best? |
| 14-01 2013 No. 13-11 | Fernau, E. Theissen N.F. Carline, S. C. Linn, P. K. Yadav Author(s) R. Baule, O. Korn, S. Saßning V. Agarwal, L. Ma, K. | Corporate Governance and the Nature of Takeover Resistance Title Which Beta is Best? On the Information Content of Option-implied Betas |
| 14-01 2013 No. 13-11 13-10 | Fernau, E. Theissen N.F. Carline, S. C. Linn, P. K. Yadav Author(s) R. Baule, O. Korn, S. Saßning V. Agarwal, L. Ma, K. Mullally M. J. Kamstra, L.A. Kramer, M.D. Levi, R. | Corporate Governance and the Nature of Takeover Resistance Title Which Beta is Best? On the Information Content of Option-implied Betas Managerial Multitasking in the Mutual Fund Industry Seasonal Asset Allocation: Evidence from |
| 14-01 2013 No. 13-11 13-10 | Fernau, E. Theissen N.F. Carline, S. C. Linn, P. K. Yadav Author(s) R. Baule, O. Korn, S. Saßning V. Agarwal, L. Ma, K. Mullally M. J. Kamstra, L.A. Kramer, M.D. Levi, R. Wermers F. Brinkmann, A. Kempf, | Corporate Governance and the Nature of Takeover Resistance Title Which Beta is Best? On the Information Content of Option-implied Betas Managerial Multitasking in the Mutual Fund Industry Seasonal Asset Allocation: Evidence from Mutual Fund Flows Forward-Looking Measures of Higher-Order Dependencies |
| 14-01 2013 No. 13-11 13-10 13-09 | Fernau, E. Theissen N.F. Carline, S. C. Linn, P. K. Yadav Author(s) R. Baule, O. Korn, S. Saßning V. Agarwal, L. Ma, K. Mullally M. J. Kamstra, L.A. Kramer, M.D. Levi, R. Wermers F. Brinkmann, A. Kempf, O. Korn G. Cici, S. Gibson, Y. Gunduz, J.J. Merrick, | Title Which Beta is Best? On the Information Content of Option-implied Betas Managerial Multitasking in the Mutual Fund Industry Seasonal Asset Allocation: Evidence from Mutual Fund Flows Forward-Looking Measures of Higher-Order Dependencies with an Application to Portfolio Selection Market Transparency and the Marking Precision of Bond |
| 14-01 2013 No. 13-11 13-10 13-09 13-08 13-07 | Fernau, E. Theissen N.F. Carline, S. C. Linn, P. K. Yadav Author(s) R. Baule, O. Korn, S. Saßning V. Agarwal, L. Ma, K. Mullally M. J. Kamstra, L.A. Kramer, M.D. Levi, R. Wermers F. Brinkmann, A. Kempf, O. Korn G. Cici, S. Gibson, Y. Gunduz, J.J. Merrick, Jr. S. Bethke, M. Gehde- | Title Which Beta is Best? On the Information Content of Option-implied Betas Managerial Multitasking in the Mutual Fund Industry Seasonal Asset Allocation: Evidence from Mutual Fund Flows Forward-Looking Measures of Higher-Order Dependencies with an Application to Portfolio Selection Market Transparency and the Marking Precision of Bond Mutual Fund Managers Investor Sentiment, Flight-to-Quality, and Corporate Bond |
| 14-01 2013 No. 13-11 13-10 13-09 13-08 13-07 | Fernau, E. Theissen N.F. Carline, S. C. Linn, P. K. Yadav Author(s) R. Baule, O. Korn, S. Saßning V. Agarwal, L. Ma, K. Mullally M. J. Kamstra, L.A. Kramer, M.D. Levi, R. Wermers F. Brinkmann, A. Kempf, O. Korn G. Cici, S. Gibson, Y. Gunduz, J.J. Merrick, Jr. S. Bethke, M. Gehde- Trapp, A. Kempf P. Schuster, M. Trapp, | Title Which Beta is Best? On the Information Content of Option-implied Betas Managerial Multitasking in the Mutual Fund Industry Seasonal Asset Allocation: Evidence from Mutual Fund Flows Forward-Looking Measures of Higher-Order Dependencies with an Application to Portfolio Selection Market Transparency and the Marking Precision of Bond Mutual Fund Managers Investor Sentiment, Flight-to-Quality, and Corporate Bond Comovement |

| | Y. Tang, B. Yang | Fund Performance |
|-------|---|---|
| 13-03 | V. Agarwal, V. Nanda, S.Ray | Institutional Investment and Intermediation in the Hedge Fund Industry |
| 13-02 | C. Andres, A. Betzer, M. Doumet, E. Theissen | Open Market Share Repurchases in Germany: A Conditional Event Study Approach |
| 13-01 | J. Gaul, E. Theissen | A Partially Linear Approach to Modelling the Dynamics of Spot and Futures Price |
| 2012 | | |
| No. | Author(s) | Title |
| 12-12 | M. Gehde-Trapp, Y. Gündüz, J. Nasev | The liquidity premium in CDS transaction prices: Do frictions matter? |
| 12-11 | Y. Wu, R. Wermers, J. Zechner | Governance and Shareholder Value in Delegated Portfolio Management: The Case of Closed-End Funds |
| 12-10 | M. Trapp, C. Wewel | Transatlantic Systemic Risk |
| 12-09 | G. Cici, A. Kempf, C. Sorhage | Do Financial Advisors Provide Tangible Benefits for Investors? Evidence from Tax-Motivated Mutual Fund Flows |
| 12-08 | S. Jank | Changes in the composition of publicly traded firms: Implications for the dividend-price ratio and return predictability |
| 12-07 | G. Cici, C. Rosenfeld | A Study of Analyst-Run Mutual Funds: The Abilities and Roles of Buy-Side Analysts |
| 12-06 | A. Kempf, A. Pütz, F. Sonnenburg | Fund Manager Duality: Impact on Performance and Investment Behavior |
| 12-05 | L. Schmidt, A. Timmermann, R. Wermers | Runs on Money Market Mutual Funds |
| 12-04 | R. Wermers | A matter of style: The causes and consequences of style drift in institutional portfolios |
| 12-03 | C. Andres, A. Betzer, I. van den Bongard, C. Haesner, E. Theissen | Dividend Announcements Reconsidered: Dividend Changes versus Dividend Surprises |
| 12-02 | C. Andres, E. Fernau, E. Theissen | Should I Stay or Should I Go? Former CEOs as Monitors |
| 12-01 | L. Andreu, A. Pütz | Choosing two business degrees versus choosing one: What does it tell about mutual fund managers' investment behavior? |
| 2011 | | |
| No. | Author(s) | Title |
| 11-16 | V. Agarwal, JP. Gómez, R. Priestley | Management Compensation and Market Timing under Portfolio Constraints |
| 11-15 | T. Dimpfl, S. Jank | Can Internet Search Queries Help to Predict Stock Market Volatility? |
| 11-14 | P. Gomber, U. Schweickert, E. Theissen | Liquidity Dynamics in an Electronic Open Limit Order Book: An Event Study Approach |

| 44.40 | | |
|--|--|---|
| 11-13 | D. Hess, S. Orbe | Irrationality or Efficiency of Macroeconomic Survey Forecasts? Implications from the Anchoring Bias Test |
| 11-12 | D. Hess, P. Immenkötter | Optimal Leverage, its Benefits, and the Business Cycle |
| 11-11 | N. Heinrichs, D. Hess, C. Homburg, M. Lorenz, S. Sievers | Extended Dividend, Cash Flow and Residual Income Valuation Models – Accounting for Deviations from Ideal Conditions |
| 11-10 | A. Kempf, O. Korn, S. Saßning | Portfolio Optimization using Forward - Looking Information |
| 11-09 | V. Agarwal, S. Ray | Determinants and Implications of Fee Changes in the Hedge Fund Industry |
| 11-08 | G. Cici, LF. Palacios | On the Use of Options by Mutual Funds: Do They Know What They Are Doing? |
| 11-07 | V. Agarwal, G. D. Gay, L. Ling | Performance inconsistency in mutual funds: An investigation of window-dressing behavior |
| 11-06 | N. Hautsch, D. Hess, D. Veredas | The Impact of Macroeconomic News on Quote Adjustments, Noise, and Informational Volatility |
| 11-05 | G. Cici | The Prevalence of the Disposition Effect in Mutual Funds' Trades |
| 11-04 | S. Jank | Mutual Fund Flows, Expected Returns and the Real Economy |
| 11-03 | G.Fellner, E.Theissen | Short Sale Constraints, Divergence of Opinion and Asset Value: Evidence from the Laboratory |
| 11-02 | S.Jank | Are There Disadvantaged Clienteles in Mutual Funds? |
| | | 3 |
| | V. Agarwal, C. Meneghetti | The Role of Hedge Funds as Primary Lenders |
| 11-01 | V. Agarwal, C. Meneghetti | · |
| 11-01 2010 | V. Agarwal, C. Meneghetti Author(s) | · |
| 11-01 2010 No. | | The Role of Hedge Funds as Primary Lenders |
| 11-01 2010 No. 10-20 | Author(s) G. Cici, S. Gibson, | The Role of Hedge Funds as Primary Lenders Title Missing the Marks? Dispersion in Corporate Bond Valuations |
| 11-01 2010 <u>No.</u> 10-20 10-19 | Author(s) G. Cici, S. Gibson, J.J. Merrick Jr. J. Hengelbrock, | The Role of Hedge Funds as Primary Lenders Title Missing the Marks? Dispersion in Corporate Bond Valuations Across Mutual Funds |
| 11-01 2010 No. 10-20 10-19 | Author(s) G. Cici, S. Gibson, J.J. Merrick Jr. J. Hengelbrock, E. Theissen, C. Westheide | Title Missing the Marks? Dispersion in Corporate Bond Valuations Across Mutual Funds Market Response to Investor Sentiment The Performance of Corporate-Bond Mutual Funds: |
| 11-01 2010 No. 10-20 10-19 10-18 | Author(s) G. Cici, S. Gibson, J.J. Merrick Jr. J. Hengelbrock, E. Theissen, C. Westheide G. Cici, S. Gibson D. Hess, D. Kreutzmann, | Title Missing the Marks? Dispersion in Corporate Bond Valuations Across Mutual Funds Market Response to Investor Sentiment The Performance of Corporate-Bond Mutual Funds: Evidence Based on Security-Level Holdings Projected Earnings Accuracy and the Profitability of Stock |
| 11-01 2010 No. 10-20 10-19 10-18 10-17 | Author(s) G. Cici, S. Gibson, J.J. Merrick Jr. J. Hengelbrock, E. Theissen, C. Westheide G. Cici, S. Gibson D. Hess, D. Kreutzmann, O. Pucker | Title Missing the Marks? Dispersion in Corporate Bond Valuations Across Mutual Funds Market Response to Investor Sentiment The Performance of Corporate-Bond Mutual Funds: Evidence Based on Security-Level Holdings Projected Earnings Accuracy and the Profitability of Stock Recommendations Sturm und Drang in Money Market Funds: When Money |
| 11-01 2010 No. 10-20 10-19 10-18 10-17 10-16 10-15 | Author(s) G. Cici, S. Gibson, J.J. Merrick Jr. J. Hengelbrock, E. Theissen, C. Westheide G. Cici, S. Gibson D. Hess, D. Kreutzmann, O. Pucker S. Jank, M. Wedow G. Cici, A. Kempf, A. | Title Missing the Marks? Dispersion in Corporate Bond Valuations Across Mutual Funds Market Response to Investor Sentiment The Performance of Corporate-Bond Mutual Funds: Evidence Based on Security-Level Holdings Projected Earnings Accuracy and the Profitability of Stock Recommendations Sturm und Drang in Money Market Funds: When Money Market Funds Cease to Be Narrow |
| 11-01 2010 No. 10-20 10-19 10-18 10-17 10-16 10-15 | Author(s) G. Cici, S. Gibson, J.J. Merrick Jr. J. Hengelbrock, E. Theissen, C. Westheide G. Cici, S. Gibson D. Hess, D. Kreutzmann, O. Pucker S. Jank, M. Wedow G. Cici, A. Kempf, A. Puetz | Title Missing the Marks? Dispersion in Corporate Bond Valuations Across Mutual Funds Market Response to Investor Sentiment The Performance of Corporate-Bond Mutual Funds: Evidence Based on Security-Level Holdings Projected Earnings Accuracy and the Profitability of Stock Recommendations Sturm und Drang in Money Market Funds: When Money Market Funds Cease to Be Narrow The Valuation of Hedge Funds' Equity Positions |
| 11-01 2010 No. 10-20 10-19 10-18 10-17 10-16 10-15 10-14 10-13 | Author(s) G. Cici, S. Gibson, J.J. Merrick Jr. J. Hengelbrock, E. Theissen, C. Westheide G. Cici, S. Gibson D. Hess, D. Kreutzmann, O. Pucker S. Jank, M. Wedow G. Cici, A. Kempf, A. Puetz J. Grammig, S. Jank | Title Missing the Marks? Dispersion in Corporate Bond Valuations Across Mutual Funds Market Response to Investor Sentiment The Performance of Corporate-Bond Mutual Funds: Evidence Based on Security-Level Holdings Projected Earnings Accuracy and the Profitability of Stock Recommendations Sturm und Drang in Money Market Funds: When Money Market Funds Cease to Be Narrow The Valuation of Hedge Funds' Equity Positions Creative Destruction and Asset Prices Purchase and Redemption Decisions of Mutual Fund |

| 10-10 | S. Frey, P. Herbst | The Influence of Buy-side Analysts on Mutual Fund Trading |
|--|--|---|
| 10-09 | V. Agarwal, W. Jiang, Y. Tang, B. Yang | Uncovering Hedge Fund Skill from the Portfolio Holdings They Hide |
| 10-08 | V. Agarwal, V. Fos, W. Jiang | Inferring Reporting Biases in Hedge Fund Databases from Hedge Fund Equity Holdings |
| 10-07 | V. Agarwal, G. Bakshi, J. Huij | Do Higher-Moment Equity Risks Explain Hedge Fund Returns? |
| 10-06 | J. Grammig, F. J. Peter | Tell-Tale Tails: A data driven approach to estimate unique market information shares |
| 10-05 | K. Drachter, A. Kempf | Höhe, Struktur und Determinanten der Managervergütung- Eine Analyse der Fondsbranche in Deutschland |
| 10-04 | J. Fang, A. Kempf, M. Trapp | Fund Manager Allocation |
| 10-03 | P. Finter, A. Niessen- Ruenzi, S. Ruenzi | The Impact of Investor Sentiment on the German Stock Market |
| 10-02 | D. Hunter, E. Kandel, S. Kandel, R. Wermers | Mutual Fund Performance Evaluation with Active Peer Benchmarks |
| 10-01 | S. Artmann, P. Finter, A. Kempf | Determinants of Expected Stock Returns: Large Sample Evidence from the German Market |
| 2009 | | |
| No. | Author(s) | Title |
| | | |
| 09-17 | E. Theissen | Price Discovery in Spot and Futures Markets: A Reconsideration |
| 09-17 | E. Theissen M. Trapp | |
| | | A Reconsideration Trading the Bond-CDS Basis – The Role of Credit Risk |
| 09-16 | M. Trapp A. Betzer, J. Gider, | A Reconsideration Trading the Bond-CDS Basis – The Role of Credit Risk and Liquidity |
| 09-16 09-15 | M. Trapp A. Betzer, J. Gider, D.Metzger, E. Theissen A. Kempf, O. Korn, | A Reconsideration Trading the Bond-CDS Basis – The Role of Credit Risk and Liquidity Strategic Trading and Trade Reporting by Corporate Insiders |
| 09-16 09-15 09-14 | M. Trapp A. Betzer, J. Gider, D.Metzger, E. Theissen A. Kempf, O. Korn, M. Uhrig-Homburg | A Reconsideration Trading the Bond-CDS Basis – The Role of Credit Risk and Liquidity Strategic Trading and Trade Reporting by Corporate Insiders The Term Structure of Illiquidity Premia Time-Varying Credit Risk and Liquidity Premia in Bond and |
| 09-16 09-15 09-14 09-13 | M. Trapp A. Betzer, J. Gider, D.Metzger, E. Theissen A. Kempf, O. Korn, M. Uhrig-Homburg W. Bühler, M. Trapp | A Reconsideration Trading the Bond-CDS Basis – The Role of Credit Risk and Liquidity Strategic Trading and Trade Reporting by Corporate Insiders The Term Structure of Illiquidity Premia Time-Varying Credit Risk and Liquidity Premia in Bond and CDS Markets Explaining the Bond-CDS Basis – The Role of Credit Risk and |
| 09-16 09-15 09-14 09-13 | M. Trapp A. Betzer, J. Gider, D.Metzger, E. Theissen A. Kempf, O. Korn, M. Uhrig-Homburg W. Bühler, M. Trapp W. Bühler, M. Trapp S. J. Taylor, P. K. Yadav, | A Reconsideration Trading the Bond-CDS Basis – The Role of Credit Risk and Liquidity Strategic Trading and Trade Reporting by Corporate Insiders The Term Structure of Illiquidity Premia Time-Varying Credit Risk and Liquidity Premia in Bond and CDS Markets Explaining the Bond-CDS Basis – The Role of Credit Risk and Liquidity |
| 09-16 09-15 09-14 09-13 09-12 | M. Trapp A. Betzer, J. Gider, D.Metzger, E. Theissen A. Kempf, O. Korn, M. Uhrig-Homburg W. Bühler, M. Trapp W. Bühler, M. Trapp S. J. Taylor, P. K. Yadav, Y. Zhang A. Kempf, C. Merkle, | A Reconsideration Trading the Bond-CDS Basis – The Role of Credit Risk and Liquidity Strategic Trading and Trade Reporting by Corporate Insiders The Term Structure of Illiquidity Premia Time-Varying Credit Risk and Liquidity Premia in Bond and CDS Markets Explaining the Bond-CDS Basis – The Role of Credit Risk and Liquidity Cross-sectional analysis of risk-neutral skewness Low Risk and High Return – Affective Attitudes and Stock |
| 09-16 09-15 09-14 09-13 09-12 09-11 | M. Trapp A. Betzer, J. Gider, D.Metzger, E. Theissen A. Kempf, O. Korn, M. Uhrig-Homburg W. Bühler, M. Trapp W. Bühler, M. Trapp S. J. Taylor, P. K. Yadav, Y. Zhang A. Kempf, C. Merkle, A. Niessen-Ruenzi V. Fotak, V. Raman, | A Reconsideration Trading the Bond-CDS Basis – The Role of Credit Risk and Liquidity Strategic Trading and Trade Reporting by Corporate Insiders The Term Structure of Illiquidity Premia Time-Varying Credit Risk and Liquidity Premia in Bond and CDS Markets Explaining the Bond-CDS Basis – The Role of Credit Risk and Liquidity Cross-sectional analysis of risk-neutral skewness Low Risk and High Return – Affective Attitudes and Stock Market Expectations |

| 09-06 | S. Frey, P. Sandas | The Impact of Iceberg Orders in Limit Order Books |
|-------|--|---|
| 09-05 | H. Beltran-Lopez, P. Giot, J. Grammig | Commonalities in the Order Book |
| 09-04 | J. Fang, S. Ruenzi | Rapid Trading bei deutschen Aktienfonds: Evidenz aus einer großen deutschen Fondsgesellschaft |
| 09-03 | A. Banegas, B. Gillen, A. Timmermann, R. Wermers | The Cross-Section of Conditional Mutual Fund Performance in European Stock Markets |
| 09-02 | J. Grammig, A. Schrimpf, M. Schuppli | Long-Horizon Consumption Risk and the Cross-Section of Returns: New Tests and International Evidence |
| 09-01 | O. Korn, P. Koziol | The Term Structure of Currency Hedge Ratios |
| 2008 | | |
| No. | Author(s) | Title |
| 08-12 | U. Bonenkamp, C. Homburg, A. Kempf | Fundamental Information in Technical Trading Strategies |
| 08-11 | O. Korn | Risk Management with Default-risky Forwards |
| 08-10 | J. Grammig, F.J. Peter | International Price Discovery in the Presence of Market Microstructure Effects |
| 08-09 | C. M. Kuhnen, A. Niessen | Public Opinion and Executive Compensation |
| 08-08 | A. Pütz, S. Ruenzi | Overconfidence among Professional Investors: Evidence from Mutual Fund Managers |
| 08-07 | P. Osthoff | What matters to SRI investors? |
| 08-06 | A. Betzer, E. Theissen | Sooner Or Later: Delays in Trade Reporting by Corporate Insiders |
| 08-05 | P. Linge, E. Theissen | Determinanten der Aktionärspräsenz auf Hauptversammlungen deutscher Aktiengesellschaften |
| 08-04 | N. Hautsch, D. Hess, C. Müller | Price Adjustment to News with Uncertain Precision |
| 08-03 | D. Hess, H. Huang, A. Niessen | How Do Commodity Futures Respond to Macroeconomic News? |
| 08-02 | R. Chakrabarti, W. Megginson, P. Yadav | Corporate Governance in India |
| 08-01 | C. Andres, E. Theissen | Setting a Fox to Keep the Geese - Does the Comply-or-Explain Principle Work? |
| 2007 | | |
| No. | Author(s) | Title |
| 07-16 | M. Bär, A. Niessen, S. Ruenzi | The Impact of Work Group Diversity on Performance: Large Sample Evidence from the Mutual Fund Industry |
| 07-15 | A. Niessen, S. Ruenzi | Political Connectedness and Firm Performance: Evidence From Germany |
| 07-14 | O. Korn | Hedging Price Risk when Payment Dates are Uncertain |
| 07-13 | A.Kempf, P. Osthoff | SRI Funds: Nomen est Omen |
| | | |

| 07-12 | J. Grammig, E. Theissen, O. Wuensche | Time and Price Impact of a Trade: A Structural Approach |
|--|---|--|
| 07-11 | V. Agarwal, J. R. Kale | On the Relative Performance of Multi-Strategy and Funds of Hedge Funds |
| 07-10 | M. Kasch-Haroutounian, E. Theissen | Competition Between Exchanges: Euronext versus Xetra |
| 07-09 | V. Agarwal, N. D. Daniel, N. Y. Naik | Do hedge funds manage their reported returns? |
| 07-08 | N. C. Brown, K. D. Wei, R. Wermers | Analyst Recommendations, Mutual Fund Herding, and Overreaction in Stock Prices |
| 07-07 | A. Betzer, E. Theissen | Insider Trading and Corporate Governance: The Case of Germany |
| 07-06 | V. Agarwal, L. Wang | Transaction Costs and Value Premium |
| 07-05 | J. Grammig, A. Schrimpf | Asset Pricing with a Reference Level of Consumption: New Evidence from the Cross-Section of Stock Returns |
| 07-04 | V. Agarwal, N.M. Boyson, N.Y. Naik | Hedge Funds for retail investors? An examination of hedged mutual funds |
| 07-03 | D. Hess, A. Niessen | The Early News Catches the Attention: On the Relative Price Impact of Similar Economic Indicators |
| 07-02 | A. Kempf, S. Ruenzi, T. Thiele | Employment Risk, Compensation Incentives and Managerial Risk Taking - Evidence from the Mutual Fund Industry - |
| 07-01 | M. Hagemeister, A. Kempf | CAPM und erwartete Renditen: Eine Untersuchung auf Basis der Erwartung von Marktteilnehmern |
| 2006 | | |
| 2000 | | |
| No. | Author(s) | Title |
| | Author(s) S. Čeljo-Hörhager, A. Niessen | Title How do Self-fulfilling Prophecies affect Financial Ratings? - An experimental study |
| No. | S. Čeljo-Hörhager, | How do Self-fulfilling Prophecies affect Financial Ratings? - An |
| No. 06-13 | S. Čeljo-Hörhager, A. Niessen R. Wermers, Y. Wu, | How do Self-fulfilling Prophecies affect Financial Ratings? - An experimental study Portfolio Performance, Discount Dynamics, and the Turnover |
| No. 06-13 06-12 | S. Čeljo-Hörhager, A. Niessen R. Wermers, Y. Wu, J. Zechner U. v. Lilienfeld-Toal, | How do Self-fulfilling Prophecies affect Financial Ratings? - An experimental study Portfolio Performance, Discount Dynamics, and the Turnover of Closed-End Fund Managers Why Managers Hold Shares of Their Firm: An Empirical |
| No. 06-13 06-12 06-11 | S. Čeljo-Hörhager, A. Niessen R. Wermers, Y. Wu, J. Zechner U. v. Lilienfeld-Toal, S. Ruenzi | How do Self-fulfilling Prophecies affect Financial Ratings? - An experimental study Portfolio Performance, Discount Dynamics, and the Turnover of Closed-End Fund Managers Why Managers Hold Shares of Their Firm: An Empirical Analysis The Effect of Socially Responsible Investing on Portfolio |
| No. 06-13 06-12 06-11 06-10 | S. Čeljo-Hörhager, A. Niessen R. Wermers, Y. Wu, J. Zechner U. v. Lilienfeld-Toal, S. Ruenzi A. Kempf, P. Osthoff R. Wermers, T. Yao, | How do Self-fulfilling Prophecies affect Financial Ratings? - An experimental study Portfolio Performance, Discount Dynamics, and the Turnover of Closed-End Fund Managers Why Managers Hold Shares of Their Firm: An Empirical Analysis The Effect of Socially Responsible Investing on Portfolio Performance Extracting Stock Selection Information from Mutual Fund |
| No. 06-13 06-12 06-11 06-10 06-09 | S. Čeljo-Hörhager, A. Niessen R. Wermers, Y. Wu, J. Zechner U. v. Lilienfeld-Toal, S. Ruenzi A. Kempf, P. Osthoff R. Wermers, T. Yao, J. Zhao | How do Self-fulfilling Prophecies affect Financial Ratings? - An experimental study Portfolio Performance, Discount Dynamics, and the Turnover of Closed-End Fund Managers Why Managers Hold Shares of Their Firm: An Empirical Analysis The Effect of Socially Responsible Investing on Portfolio Performance Extracting Stock Selection Information from Mutual Fund holdings: An Efficient Aggregation Approach The Poole Analysis in the New Open Economy |
| No. 06-13 06-12 06-11 06-10 06-09 06-08 | S. Čeljo-Hörhager, A. Niessen R. Wermers, Y. Wu, J. Zechner U. v. Lilienfeld-Toal, S. Ruenzi A. Kempf, P. Osthoff R. Wermers, T. Yao, J. Zhao M. Hoffmann, B. Kempa K. Drachter, A. Kempf, | How do Self-fulfilling Prophecies affect Financial Ratings? - An experimental study Portfolio Performance, Discount Dynamics, and the Turnover of Closed-End Fund Managers Why Managers Hold Shares of Their Firm: An Empirical Analysis The Effect of Socially Responsible Investing on Portfolio Performance Extracting Stock Selection Information from Mutual Fund holdings: An Efficient Aggregation Approach The Poole Analysis in the New Open Economy Macroeconomic Framework Decision Processes in German Mutual Fund Companies: |
| No. 06-13 06-12 06-11 06-10 06-09 06-08 06-07 | S. Čeljo-Hörhager, A. Niessen R. Wermers, Y. Wu, J. Zechner U. v. Lilienfeld-Toal, S. Ruenzi A. Kempf, P. Osthoff R. Wermers, T. Yao, J. Zhao M. Hoffmann, B. Kempa K. Drachter, A. Kempf, M. Wagner J.P. Krahnen, F.A. | How do Self-fulfilling Prophecies affect Financial Ratings? - An experimental study Portfolio Performance, Discount Dynamics, and the Turnover of Closed-End Fund Managers Why Managers Hold Shares of Their Firm: An Empirical Analysis The Effect of Socially Responsible Investing on Portfolio Performance Extracting Stock Selection Information from Mutual Fund holdings: An Efficient Aggregation Approach The Poole Analysis in the New Open Economy Macroeconomic Framework Decision Processes in German Mutual Fund Companies: Evidence from a Telephone Survey Investment Performance and Market Share: A Study of the |
| No. 06-13 06-12 06-11 06-10 06-09 06-08 06-07 06-06 | S. Čeljo-Hörhager, A. Niessen R. Wermers, Y. Wu, J. Zechner U. v. Lilienfeld-Toal, S. Ruenzi A. Kempf, P. Osthoff R. Wermers, T. Yao, J. Zhao M. Hoffmann, B. Kempa K. Drachter, A. Kempf, M. Wagner J.P. Krahnen, F.A. Schmid, E. Theissen | How do Self-fulfilling Prophecies affect Financial Ratings? - An experimental study Portfolio Performance, Discount Dynamics, and the Turnover of Closed-End Fund Managers Why Managers Hold Shares of Their Firm: An Empirical Analysis The Effect of Socially Responsible Investing on Portfolio Performance Extracting Stock Selection Information from Mutual Fund holdings: An Efficient Aggregation Approach The Poole Analysis in the New Open Economy Macroeconomic Framework Decision Processes in German Mutual Fund Companies: Evidence from a Telephone Survey Investment Performance and Market Share: A Study of the German Mutual Fund Industry On the Usability of Synthetic Measures of Mutual Fund Net- |
| No. 06-13 06-12 06-11 06-10 06-09 06-08 06-07 06-06 06-05 | S. Čeljo-Hörhager, A. Niessen R. Wermers, Y. Wu, J. Zechner U. v. Lilienfeld-Toal, S. Ruenzi A. Kempf, P. Osthoff R. Wermers, T. Yao, J. Zhao M. Hoffmann, B. Kempa K. Drachter, A. Kempf, M. Wagner J.P. Krahnen, F.A. Schmid, E. Theissen S. Ber, S. Ruenzi | How do Self-fulfilling Prophecies affect Financial Ratings? - An experimental study Portfolio Performance, Discount Dynamics, and the Turnover of Closed-End Fund Managers Why Managers Hold Shares of Their Firm: An Empirical Analysis The Effect of Socially Responsible Investing on Portfolio Performance Extracting Stock Selection Information from Mutual Fund holdings: An Efficient Aggregation Approach The Poole Analysis in the New Open Economy Macroeconomic Framework Decision Processes in German Mutual Fund Companies: Evidence from a Telephone Survey Investment Performance and Market Share: A Study of the German Mutual Fund Industry On the Usability of Synthetic Measures of Mutual Fund Net-Flows |

| 06-01 | A. Niessen, S. Ruenzi | Sex Matters: Gender Differences in a Professional Setting |
|-------|--|--|
| 2005 | | |
| No. | Author(s) | Title |
| 05-16 | E. Theissen | An Analysis of Private Investors' Stock Market Return Forecasts |
| 05-15 | T. Foucault, S. Moinas, E. Theissen | Does Anonymity Matter in Electronic Limit Order Markets |
| 05-14 | R. Kosowski, A. Timmermann, R. Wermers, H. White | Can Mutual Fund "Stars" Really Pick Stocks? New Evidence from a Bootstrap Analysis |
| 05-13 | D. Avramov, R. Wermers | Investing in Mutual Funds when Returns are Predictable |
| 05-12 | K. Griese, A. Kempf | Liquiditätsdynamik am deutschen Aktienmarkt |
| 05-11 | S. Ber, A. Kempf, S. Ruenzi | Determinanten der Mittelzuflüsse bei deutschen Aktienfonds |
| 05-10 | M. Bär, A. Kempf, S. Ruenzi | Is a Team Different From the Sum of Its Parts? Evidence from Mutual Fund Managers |
| 05-09 | M. Hoffmann | Saving, Investment and the Net Foreign Asset Position |
| 05-08 | S. Ruenzi | Mutual Fund Growth in Standard and Specialist Market Segments |
| 05-07 | A. Kempf, S. Ruenzi | Status Quo Bias and the Number of Alternatives - An Empirical Illustration from the Mutual Fund Industry |
| 05-06 | J. Grammig, E. Theissen | Is Best Really Better? Internalization of Orders in an Open Limit Order Book |
| 05-05 | H. Beltran-Lopez, J. Grammig, A.J. Menkveld | Limit order books and trade informativeness |
| 05-04 | M. Hoffmann | Compensating Wages under different Exchange rate Regimes |
| 05-03 | M. Hoffmann | Fixed versus Flexible Exchange Rates: Evidence from Developing Countries |
| 05-02 | A. Kempf, C. Memmel | Estimating the Global Minimum Variance Portfolio |
| 05-01 | S. Frey, J. Grammig | Liquidity supply and adverse selection in a pure limit order book market |
| 2004 | | |
| No. | Author(s) | Title |
| 04-10 | N. Hautsch, D. Hess | Bayesian Learning in Financial Markets – Testing for the Relevance of Information Precision in Price Discovery |
| 04-09 | A. Kempf, K. Kreuzberg | Portfolio Disclosure, Portfolio Selection and Mutual Fund Performance Evaluation |
| 04-08 | N.F. Carline, S.C. Linn, P.K. Yadav | Operating performance changes associated with corporate mergers and the role of corporate governance |
| 04-07 | J.J. Merrick, Jr., N.Y. Naik, P.K. Yadav | Strategic Trading Behaviour and Price Distortion in a Manipulated Market: Anatomy of a Squeeze |
| 04-06 | N.Y. Naik, P.K. Yadav | Trading Costs of Public Investors with Obligatory and Voluntary Market-Making: Evidence from Market Reforms |
| 04-05 | A. Kempf, S. Ruenzi | Family Matters: Rankings Within Fund Families and Fund Inflows |

| 04-04 | V. Agarwal, N.D. Daniel, N.Y. Naik | Role of Managerial Incentives and Discretion in Hedge Fund Performance |
|-------|--|--|
| 04-03 | V. Agarwal, W.H. Fung, J.C. Loon, N.Y. Naik | Risk and Return in Convertible Arbitrage: Evidence from the Convertible Bond Market |
| 04-02 | A. Kempf, S. Ruenzi | Tournaments in Mutual Fund Families |
| 04-01 | I. Chowdhury, M. Hoffmann, A. Schabert | Inflation Dynamics and the Cost Channel of Monetary Transmission |

